

The World's Largest Open Access Agricultural & Applied Economics Digital Library

# This document is discoverable and free to researchers across the globe due to the work of AgEcon Search.

Help ensure our sustainability.

Give to AgEcon Search

AgEcon Search
<a href="http://ageconsearch.umn.edu">http://ageconsearch.umn.edu</a>
<a href="mailto:aesearch@umn.edu">aesearch@umn.edu</a>

Papers downloaded from **AgEcon Search** may be used for non-commercial purposes and personal study only. No other use, including posting to another Internet site, is permitted without permission from the copyright owner (not AgEcon Search), or as allowed under the provisions of Fair Use, U.S. Copyright Act, Title 17 U.S.C.



#### FCND DISCUSSION PAPER NO. 104

## AN EVALUATION OF THE IMPACT OF PROGRESA ON PRESCHOOL CHILD HEIGHT

Jere R. Behrman and John Hoddinott

**Food Consumption and Nutrition Division** 

International Food Policy Research Institute 2033 K Street, N.W. Washington, D.C. 20006 U.S.A. (202) 862–5600 Fax: (202) 467–4439

March 2001

FCND Discussion Papers contain preliminary material and research results, and are circulated prior to a full peer review in order to stimulate discussion and critical comment. It is expected that most Discussion Papers will eventually be published in some other form, and that their content may also be revised.

#### **ABSTRACT**

The nutrition of preschool children is of considerable interest not only because of concern over their immediate welfare, but also because their nutrition in this formative stage of life is widely perceived to have substantial persistent impact on their physical and mental development and on their health status as adults. Children's physical and mental development shapes their later lives by affecting their schooling success and post-schooling productivity. Improving the nutritional status of currently malnourished preschoolers may, therefore, have important payoffs over the long term. Within rural Mexico, stunting, or short height relative to standards established for healthy populations, is the major form of protein-energy malnutrition (PEM). Low weight for height, or wasting, is much less of a problem. But stunting is symptomatic of longer-term effects of early childhood malnutrition.

One of the major components of the PROGRESA program has been directed toward improving the nutritional status of children in poor rural communities in Mexico. Cross-sectional comparisons of height for children who received PROGRESA treatment versus others who were in PROGRESA-eligible households but who did not receive treatment suggest no positive effect of PROGRESA, either on average child height or on the proportion of children who are stunted, i.e., more than two standard deviations below recognized norms. But these comparisons may be misleading because of the failure to control for unobserved child, parental and household, and market and community characteristics that may be correlated with children receiving the PROGRESA treatment,

or because of the failure to control for systematic initial differences. For example, on average, the children in the control sample tended to have better anthropometric status than children in the treatment sample.

The preferred estimates used in this study control for these factors. PROGRESA treatment is represented by those who reportedly received nutritional supplements in the treatment group (less than 60 percent of children in the treatment group) for children in the critical age range of 12 to 36 months. These estimates find significant effects of receiving PROGRESA treatment in increasing child growth and reducing the probability of child stunting. These estimates imply an increase of about one-sixth in mean growth per year for these children, and perhaps somewhat greater for children from poorer households and poorer communities but whose household heads are more educated. This is a potentially important effect: under the assumptions that (1) there is strong persistence of changes in small children's anthropometric development so that the percentage changes for adults equal those (are half of those) that we estimate for children and (2) that adult anthropometric-earnings relations from elsewhere in Latin America apply to the labor markets in which these children will be working as adults, the impact from this effect alone would be a 2.9 percent (1.4 percent) increase in lifetime earnings. In addition, there are likely to be other effects through increased cognitive development, increased schooling, and lowered age of completing given levels of schooling through starting when younger and passing successfully grades at a higher rate. While these estimates remain fairly speculative, they suggest that PROGRESA may have substantial effects on lifetime productivity and earnings of preschool children in poor households.

### **CONTENTS**

Ackno	owledgments	vi
1. Intro	oduction	1
2. PRO	OGRESA Intervention	4
3. Con	nceptual Framework	9
4. Data	a	14
5. Esti	mates of PROGRESA Impact	24
	Estimated Impact on Child Height	24 27 33
6. Lon	ger-Term Impact	45
7. Sun	nmary and Conclusions	48
Refere	ences	51
	TABLES	
1	Nutrition status indicators for children aged 12-36 months in households eligible for PROGRESA in INSP 1998 August survey	2
2	Percentage of children appearing in both 1998 and 1999 survey rounds, by age, state, and household eligibility for PROGRESA	16
3	Mean height-for-age Z-scores of children measured only in 1998 and measured in 1998 and 1999, by age group	17
4	INSP longitudinal sample for 1998 and 1999: Numbers of children less than 60 months of age in August 1998 in PROGRESA control and treatment communities who are eligible and not eligible for PROGRESA	18

5	Proportion of children in eligible treatment households who received nutrition supplements and number of children in INSP sample of eligible households, by age groups	20
6	Means and standard deviations for children eligible for PROGRESA less than 60 months of age and variables for which the means differ significantly between eligible treatment and control households in August 1998 for all children less than 60 months and for those 12-36 months	21
7	OLS estimates of impact of treatment on height in 1999 for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA.	25
8	Individual child fixed effects estimates of impact of treatment on height for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA	28
9	Relation between estimated individual child growth fixed effects from Table 8 and observed fixed child, household and parental, price, community, and state characteristics for children 12-36 months in August 1998 and eligible for PROGRESA	31
10	Individual child fixed effects estimates of impact of treatment on height for children in August 1998 who are in households eligible for PROGRESA for alternative age ranges	34
11a	Means, standard deviations, and correlations for four alternative representations of PROGRESA treatment for children in eligible households who were 12-36 months of age in August 1998	37
11b	Individual child fixed effects estimates of impact of treatment on height for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA for alternative representations of PROGRESA	38
12	Summary of interaction effects between receiving treatment and child, household, community, and state characteristics	40
13	Logit estimates of impact of treatment on stunting in 1999 for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA	44
14	Individual child fixed effects logit estimates of impact of treatment on stunting for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA	44

vi

#### ACKNOWLEDGMENTS

The authors thank Ryan Washburn for excellent research assistance, and David Coady, Marie Ruel, and Emmanuel Skoufias for help in understanding aspects of PROGRESA that are related to child nutrition and growth. We are most grateful to the *Instituto Nacional de Salud Pública* who collected the anthropometric data used here and have generously shared them with us. We are also pleased to thank Daniel Hernandez, Maria del Carmen Huerta, Monica Orozco, and Patricia Muniz for their assistance. The authors alone are responsible for all the contents and interpretations in this paper.

Jere R. Behrman University of Pennsylvania

John Hoddinott International Food Policy Research Institute

#### 1. INTRODUCTION

The nutrition of preschool children is of considerable interest not only because of concern over their immediate welfare, but also because their nutrition in this formative stage of life is widely perceived to have substantial persistent impact on their physical and mental development and on their health status as adults. Their physical and mental development, in turn, shapes their lifetime options by affecting their schooling success and post-schooling productivity. Improving the nutritional status of malnourished infants and small children may, therefore, have important payoffs over the long term. <sup>1</sup>

Malnutrition can take many forms. Longer-run basic macro or protein-energy malnutrition (PEM) usually is manifested in stunting, i.e., being short for one's age and sex relative to standards established for healthy populations. Shorter-run PEM often is measured by wasting (low weight-for-height), low weight for one's age and sex, or a low body-mass-index (BMI, weight-per-height squared). Micronutrient deficiencies can be identified by various observational and clinical measures, depending on the nature of the deficiency. Blood tests, for example, can identify iron deficiencies.

Table 1 summarizes information on some of the major indicators of PEM for children ages 12–36 months who are in households eligible for PROGRESA in August

\_

<sup>&</sup>lt;sup>1</sup> A number of aspects of the literature on child health and nutrition and on the impact of health and nutrition on productivity in school and subsequent to school are discussed in Adair (1999), Alderman et al. (2000), Behrman (1993, 1996), Behrman and Deolalikar (1988), Glewwe, Jacoby, and King (2000), Golden (1994), Grantham-McGregor et al. (1997), Grantham-McGregor, Fernald, and Sethuraman (1999), Johnston et al. (1987), Leslie and Jamison (1990), Martorell (1995, 1999), Martorell, Rivera, and Kaplowitz (1989), Martorell, Khan, and Schroeder (1994), Pollitt (1990), Pollitt et al. (1993), Strauss and Thomas (1995, 1998), and Young (1995). Also, see Section 6 below.

1998 (before the infant and small child nutrition component of the program was initiated in the sample from which these data are drawn).<sup>2</sup> The first three rows give information on Z-scores for height-for-age, weight-for-age, and weight-for-height. Z-scores give the number of standard deviations from the mean score for the same anthropometric measure conditional on age in a healthy population—in particular the NCHS/WHO standards based on a healthy U.S. population that are widely used in the literature. The last two rows give the proportions stunted and wasted, defined here as being more than two standard deviations below the mean for the standards.

Table 1—Nutrition status indicators for children aged 12–36 months in households eligible for PROGRESA in INSP 1998 August survey

Nutrition Status Indicator	Mean	Standard Deviation	95% Confide	ence interval	Sample size
Z score for height-for-age <sup>a</sup>	-1.76	1.34	-1.91	-1.62	316
Z score for weight-for-age <sup>a</sup>	-1.40	1.16	-1.53	-1.27	316
Z score for weight-for-height <sup>a</sup>	-0.41	1.09	-0.53	-0.29	316
Proportion stunted <sup>b</sup>	0.44	0.50	0.38	0.49	320
Proportion wasted <sup>b</sup>	0.06	0.23	0.03	0.08	320

<sup>&</sup>lt;sup>a</sup> The Z scores give the number of standard deviations from the means for healthy children as defined in the NCS standards for the United States. The sample sizes differ slightly between the z scores and the proportions stunted and wasted because of a few outliers with very high z scores for height and weight (i.e., more that six standard deviations above the mean) that are excluded from these calculations for the z scores.

b These are the proportions that are more than two standard deviations below the norms (i.e., with Z scores < -2).

<sup>&</sup>lt;sup>2</sup> The data are discussed below. As also is discussed below, we focus on this group of children in this study.

Note that for the PROGRESA target population in this age range, the dominant form of PEM is stunting rather than wasting. On average, these children are 1.76 standard deviations below the norm in terms of height-for-age, and 44 percent are more than 2.0 standard deviations below the norm. By contrast, they average only 0.41 standard deviations below the norm with regard to weight-for-height, and only 6 percent are more than 2.0 standard deviations below the norm (and thus wasted). Since there is a substantial incidence of stunting with its potentially damaging long-run effects, but not much wasting,<sup>3</sup> this report focuses on the impact of PROGRESA on stunting.<sup>4</sup>

After this introduction, Section 2 summarizes important aspects of the PROGRESA intervention in general and with regard to infant and small child nutrition. Section 3 presents a conceptual framework that guides the empirical exploration. Section 4 presents the data and considers bivariate patterns in nutritional status and comparisons between the treatment and control communities. Section 5 discusses our basic estimates of the impact of PROGRESA on child growth and stunting and considers some variants of those estimates. Section 6 uses our basic estimates to simulate the impact on lifetime productivity and income. Finally, Section 7 summarizes and concludes.

-

<sup>&</sup>lt;sup>3</sup> These same patterns, of course, are reported in the INSP (1998) summary of these data. They also are common in many other populations in the region, including for example, those in Central America by Johnston, *et al.* (1987), Martorell (1995, 1999), Martorell, Khan, and Schroeder (1994), Martorell, Rivera, and Kaplowitz (1989), and others.

<sup>&</sup>lt;sup>4</sup> We have undertaken parallel explorations in the determinants of wasting and anemia. These do not indicate that PROGRESA has had significant impact. But nutritional status is much better with regard to these indicators in the population of interest, so the lack of a significant impact of PROGRESA on them is not a matter of the same concern as the question of the impact on stunting.

#### 2. PROGRESA INTERVENTION

PROGRESA aims to provide support for families living in extreme poverty in small communities in rural Mexico. To broaden opportunities and capabilities to attain higher levels of well-being, the program attempts to raise living standards by improving opportunities to gain access to education, health, and food. Specifically, PROGRESA has the following objectives:<sup>5</sup>

- A. To improve substantially the conditions of education, health, and nutrition of poor families, particularly children and their mothers, by providing sufficient quality services in the areas of education and health, as well as providing monetary assistance and nutrition supplements.
- B. To integrate these actions so that educational achievement is not affected by poor health or malnutrition in children and young people, or because they carry out work that makes school attendance difficult.
- C. To ensure that households have sufficient means and resources available so that their children can complete their basic education.
- D. To encourage the responsibility and active participation of parents and all family members in improving the education, health and nutrition of children and young people.
- E. To promote community participation and support for the actions of PROGRESA, so that educational and health services benefit all families in the localities where it operates, as well as uniting and promoting community

\_

<sup>&</sup>lt;sup>5</sup> The statements regarding the specific objectives and components of PROGRESA are based directly on Coady (1999) and Skoufias, Davis, and Behrman (1999).

efforts and initiatives in actions that are similar or complementary to the program.

PROGRESA comprises three components that are closely linked to each other:

- (1) Educational grants to facilitate and encourage the educational aspirations of children and young people by fostering their enrollment and regular school attendance, and promoting parents' appreciation of the advantages of their children's education. At the same time, the quality of education is improved.
- (2) Basic health care for all members of the family and strengthening the quality of services as well as reorienting individuals and health services toward taking preventive actions toward health care and nutrition.
- (3) Monetary transfers and nutritional supplements to improve the food consumption and nutritional state of poor families, emphasizing that the purpose of this is to improve the family's food intake, particularly of children and women, who are generally the household members who suffer most from nutritional deficiencies.

Each of these components can be viewed as a form of human capital that enters directly into individual well-being, e.g., enabling one to contribute to and participate in society, but also indirectly in determining an individual's productivity and thus income-earning potential. The nature of the education-health-nutrition nexus is often seen as the root of the vicious cycle of poverty, whereby children born into poor families disproportionately experience health and nutritional problems, which diminish their

potential for benefiting from whatever education they receive. Public action is necessary if this vicious cycle is to be transformed into a virtuous one.

This paper measures the impact of PROGRESA on growth of small children who, as noted in Table 1, often suffer long-run malnourishment and stunting in the target population with possible long-term negative effects on their health and productivity.

There are at least four pathways by which participation in PROGRESA might affect child growth:

(1) *Cash Transfers*. Some monetary transfers are motivated, as noted, by the desire to improve nutrition, particularly that of young children and mothers. There has been considerable controversy in the literature over the extent to which increased income translates into increased nutrient consumption. Estimates for the PROGRESA sample indicate that a 10 percent increase in income translates into a 3 to 4.5 percent increase in caloric availability, with some of the rest of the incremental income used to purchase "better" food, i.e., richer in micronutrients (Hoddinott and Skoufias 2000). There remains, of course, the question of intrahousehold distribution and how much of such increments in nutrients goes to small children. While there is no direct evidence on the intrahousehold distribution of nutrients in the PROGRESA

<sup>&</sup>lt;sup>6</sup> See, for example, Alderman (1986, 1993), Behrman and Deolalikar (1987, 1988), Behrman, Foster and Rosenzweig (1997), Bouis (1994), Bouis and Haddad (1992), Strauss and Thomas (1995, 1998) and Subramanian and Deaton (1996).

population, studies on other poor populations have concluded that larger shares of resources that go to mothers are directed toward child health and nutrition than of resources directed to fathers, which is one reason, of course, why PROGRESA directs resources to mothers.<sup>7</sup>

- (2) *Growth Monitoring*. A prerequisite for receiving the nutritional supplements
   (see (4) below) is ongoing growth monitoring of the relevant children.
   Conventional wisdom holds that there is a high payoff to such monitoring because it increases substantially the probability that parents (or other caretakers) become aware of PEM and some other nutritional problems being experienced by their children at an earlier stage.
- (3) Participation in Meetings with Health Care Specialists. PROGRESA participants are required to attend regular meetings at which health and nutrition issues and practices are discussed, among other things. These sessions are conducted by physicians and nurses trained in these specific topics (Rivera et al. 2000). If these meetings improve knowledge and practices related to child nutrition and health, they may increase child growth.
- (4) *Nutritional Supplements*. The nutritional component of PROGRESA includes the provision of food supplements to pregnant and lactating women and to children between the ages of four months and two years and to children

<sup>7</sup> See, for example, Alderman et al. (1995), Behrman (1997), Haddad and Hoddinott (1994), Haddad, Hoddinott, and Alderman (1997), Hoddinott and Haddad (1995), Strauss and Thomas (1995), and Thomas (1990, 1993, 1994).

-

between two and five years if any signs of malnutrition are detected. The food supplements are produced at a production plant devoted solely to this task and then distributed to health centers through DICONSA, an operational arm of the Ministry of Social Development (SEDESOL) and the largest distributor of food in rural areas. There are about 18,000 DICONSA stores in rural areas. Mothers visit the clinic at least once a month (more if they are pregnant or have small children) to pick up six packets of supplements per child per month with each pack containing five doses, enough for one dose per day. The supplements constitute 20 percent of caloric requirements and 100 percent of all necessary micronutrients. Rosado et al. (1999) found that these supplements had presentational and flavor characteristics that resulted in high levels of acceptability and intake and that intake of the supplement averaged about 87 percent of the ration.

While this paper attempts to estimate the total impact of PROGRESA, it cannot identify the separate impact of these four possible pathways. For our preferred representation of having PROGRESA treatment, we use information on whether children have received supplements as well as whether they are in treatment households. The estimates presented in Section 5 indicate that this measure of having "received treatment" is more consistent with reported child growth than is "listed treatment" (see Section 4).

#### 3. CONCEPTUAL FRAMEWORK

This study assumes that parental decisions to devote resources to improving the health and nutritional status of their children are motivated both by immediate concern about their children's welfare and longer-term concern about investing in the human capital of their children (see Becker 1967). Parents presumably have this concern because of some mixture of altruistic concern about their children and the possibility of sharing in some of the returns from these human capital investments, e.g., being provided for in old age. Parents may not have identical preferences regarding the use of family resources, but may engage in (perhaps implicit) bargaining about such allocations, in which the strength of the bargaining position of individuals may depend on their access to resources, including those provided by social networks and policies. Decisions that parents make about devoting resources to the children's nutrition and health—whether arrived at through bargaining or some other mechanism—are under constraints imposed by resources that parents have and expect to have in the future, prices in markets they face and expect to face in the future, and community resources that they have access to and expect to have access to in the future. Expectations are important because, for example, the return to investments in the health and nutrition of small children will not all be realized for many years, and the extent of those returns will depend upon what will be the value in labor and other markets of increased productivity in the future.

These concerns can be formalized in a tractable manner by assuming maximization of intertemporally separable preference functions subject to intrahousehold

decision rules, resource, market, and community constraints. This process leads to reduced-form dynamic decision rules or "demand" relations that give some behavioral outcome in the current period as dependent on all predetermined (from the point of view of the entity—family, or whatever—making the decisions) prices and resources and on the parameters in the underlying production functions and preferences.

On a general level, demand functions can be written as a vector of behavioral outcomes (Z) dependent on a vector of prices broadly-defined (P) and a vector of resources (R)—with the relevant prices and resources depending on what entity is the demand function. If there are uncertainties regarding relevant future prices, policies, and shocks, the characteristics known at the time of the decision of interest regarding the distribution of those outcomes should be included instead of their realized values. A linear approximation to the demand function for a family facing prices PF and with resources RF and a vector of stochastic terms (V) is

$$(1A) Z_f = b_{PF}PF + b_{RF}RF + V,$$

where the b's are the parameters to be estimated and indicate the impact of the variables for which they are coefficients on the demands for  $Z_f$ . The stochastic term in each relation includes all the effects of all the stochastic terms in all of the production activities in which the family is engaged, plus perhaps other chance events. One of the behavioral outcomes determined in this process is children's height. Relevant resources include characteristics of each individual in the household (e.g., innate robustness of the child under consideration), characteristics of the household (e.g., overall household resources

and household size), characteristics of the community (e.g., nature of household services), and past shocks (e.g., a child having had some contagious diseases). Both prices and resources may be observed or unobserved in the data, so it is useful to indicate that distinction (using superscripts "o" and "u"). There is one such demand relation (or one element in the vector  $Z_f$ ) for every behavioral outcome of the family, including all human resource investments and all behavioral inputs that affect human resource investments through production relations. Each of these demand relations conceptually includes the same identical right-side predetermined variables, reflecting that there may be important cross-effects (e.g., the nature of health services may affect schooling demand, and vice versa) so that a nutrition or health aspect of *PROGRESA* may affect educational (and other) outcomes, and vice versa. That means that any predetermined variable that affects any one behavioral outcome may affect any or all other behavioral outcomes.

For the particular human resource of interest in this paper, the health/height of the *i*th child (*Hijt*) in the *t*th period, this relation can be written as

 $(1B) H_{ift} = b_{PFO}PF_t^O + b_{PFU}PF_t^U + b_{RFO}RF_t^O + b_{RFU}RF_t^U + cPROG_t + V_t$ , where PROG refers to PROGRESA, one of the possible household resources, that has been singled out because of its central interest for this paper, and the subscript "t" on the right-side variables refers to the vectors of past, current, and expected future values of the respective variables as of time t. We want to obtain an estimate of the impact of PROG on child health/height, i.e., a good estimate of the parameter "c." The basic estimation

problem is that there are likely to be many unobserved variables that affect child height within this framework and that may be correlated with whether a particular child in a particular household in a particular community participates in PROGRESA. For example, a household with access to nutritional supplements through PROGRESA may be more likely to take advantage of the program if the child is innately less healthy, if the parents have greater concern about their children's welfare and future prospects, if the parents perceive that the future returns to human capital investments are higher, if there are not good market alternatives or social services through which human capital investments in children could be financed, or if the local environment is relatively unhealthy. As a result, if there is not control for such factors, the estimated "c" will be contaminated by omitted variable bias and may differ substantially and possibly even in sign from the true value.

The estimation strategy used in this study controls for all time invariant unobserved child, parental and household, and market and community characteristics through fixed effects estimates. Given that there are two rounds of INSP data on child nutrition—one before access to the PROGRESA nutrition components was initiated and one about a year after they were initiated, this is equivalent to estimating the first difference of relation (1*B*). Under the assumption that all the unobserved factors are fixed, this leads to

$$(1C) H_{ift} - H_{ift-1} = b_{PFO}(PF_t^O - P_{ft-1}^O) + b_{RFO}(RF_t^O - RF_{t-1}^O) + cPROG_t + V_t - V_{t-1}.$$

The dependent variable is child growth between the interview rounds. The first two rightside variables are the changes in the transitory components of prices and of resources. These are changes in the transitory components because the permanent or longer-term components are basically fixed over time, so they are differenced out in (1C) (or—equivalently—controlled for in the fixed effects estimates). It is not clear that these transitory components are likely to have a significant effect on longer-run investments such as in children's human capital, but that is an empirical question explored below. The coefficient of  $PROG_t$  is the estimated impact of PROGRESA on child growth (recall that  $PROG_{t-1}$  is zero because the baseline survey was supposed to be conducted prior to the intervention commencing).  $V_t - V_{t-1}$  is the difference in stochastic shocks, and does not cause any biases. Under the assumptions to obtain (1C), the estimates obtained of the impact of PROGRESA are unbiased.

Thus, this estimation strategy focuses on the fixed effects equivalent of relation (1*C*) as preferred estimates. But (1*B*) is also estimated to explore whether control for unobserved fixed factors makes any difference—and finds that it makes a considerable difference. Of further note is that the logic of the model underlying (1*C*) includes the possibility that the PROGRESA impact on child growth may differ, depending on the nature of the child (e.g., be bigger for innately more sickly children), the circumstances of the household (e.g., be bigger for families that are poorer and have less access to markets and other public services or for families with more education—which enables them to exploit more quickly and more effectively the new options available because of PROGRESA). Therefore this paper explores the possibility that in (1*C*) the parameter "*c*" depends on individual child, parental and household, and community characteristics. Section 5 presents these estimates.

#### 4. DATA

The basic data source for the dependent variables of interest and for some of the right-side variables is the two rounds of data collected by the *Instituto Nacional de Salud Publica* (INSP) in August–September 1998 and October–December 1999. Survey design, sampling, sample size calculations, and other aspects of the collection of these data are summarized in INSP (1998). The 1998 survey collected basic anthropometric information on approximately 4,000 children residing in six states, Guerrero, Hidalgo, Puebla, Querétero, San Luis Potosi, and Veracruz.

For this analysis, it was necessary to have usable data on children measured in both 1998 and 1999. In this context, usable means that it must be possible to link these children to their mothers (e.g., to determine whether the impact of PROGRESA varies across children whose mothers have differing levels of education), their households (e.g., to determine if the impact of PROGRESA varies across indigenous and nonindigenous parents) and their communities (e.g., to control for food prices as possible determinants of child health). Meeting all these requirements results in a data set with substantially fewer observations than that found in the raw INSP data sets.

Rivera et al. (2000) report that the 1998 survey round was designed to contain approximately 4,480 children. However, 831 children reside in localities not surveyed as part of the ENCEL surveys, which provide much of this study's information on household characteristics. A further 380 children reside in households not in the ENCEL surveys, although they do reside in localities that were part of the ENCEL surveys. It

15

proved impossible to match 586 children to their mothers and a further 83 children had ages in excess of 60 months or could not be uniquely identified within the household. This left a sample of 2,597 children measured in 1998 with usable data.

Next, the data were merged with the data collected in 1999. The 1999 survey round collected anthropometric data on approximately 5,000 children. However, according to INSP records (after dropping observations with missing heights, where location was miscoded, and where the child resided in Morelos state, which is not part of the ENCEL surveys), only 1,639 children measured in 1998 were remeasured in 1999. These data were merged into the 1998 sample using information on state, municipality, locality, the household identifier number (folio), and child's birth date to ensure that a child measured in 1998 was correctly matched to data collected in 1999. This produced a sample of 693 children.

Accordingly, the next step was to check why so many children were "lost" between 1998 and 1999. To start, each record of children appearing in 1998 but not in 1999 was examined. The working assumption was that information on child's residence (state, municipality, locality, and folio) was entered correctly, but there were errors in entering birth dates. This proved to be most informative. Three errors emerged: (1) minor errors in entering birth dates. For example, a birth date was recorded as September 1, 1995, in the 1998 round and September 2, 1995, in 1999; (2) reversing dates and months. For example, a child recorded as being born on September 1, 1995 (09/01/95) in 1998 is recorded as being born on January 9, 1995 (01/09/95) in the 1999 round; and (3) differences in the year of birth recorded in different survey rounds. For example, a

child might have a birth date recorded as September 1, 1995 in 1998 and September 1, 1996 in 1999. Such cases became obvious upon examination of the height-for-age Z-scores across the two survey rounds. This work produced an additional 220 children for this merged data set.

In addition, some exploration was undertaken regarding the remaining lost children. It appears that a major reason for the loss of these children is age and location related. For example, of the 1,089 children aged over 36 months at the time of the 1998 survey round, only 182 were remeasured in 1999. Table 2 summarizes the percentage of children appearing in the 1998. Further, remeasured rates are far less common in

Table 2—Percentage of children appearing in both 1998 and 1999 survey rounds, by age, state, and household eligibility for PROGRESA

		Chi Squared Test for independence
	Percentage of children	of likelihood of remeasurement by
	measured in 1998 and 1999	selected categories
Child age in 1998		
< 6 months	65.8%	
6-12 months	66.2	
12-18 months	38.6	
18-24 months	47.6	
24-36 months	39.6	
>36 months	16.7	
		Chi squared (d.f.=5) = $378.3*$
Child resides in		
Guerrero	33.9%	
Hidalgo	35.8	
Puebla	38.7	
Queretaro	41.0	
San Luis Potosi	39.4	
Veracruz	25.5	
		Chi squared (d.f.=5) = $30.3*$
Child resides in household		
Eligible for PROGRES A	21.5%	
Not eligible for	26.1	
PROGRESA		
		Chi squared $(d.f.=1) = 0.6$

Note: \* differences are significant at the 5 percent level.

Veracruz than in other states. Although the likelihood of being remeasured does not vary by PROGRESA eligibility, it does vary by child height-for-age. In particular, as Table 3 indicates, children less than 36 months who were measured in both 1998 and 1999 typically had poorer height-for-age Z-scores than children who were not remeasured. This feature of the data further strengthens the importance of an estimation strategy that controls for such characteristics.

Table 3—Mean height-for-age Z-scores of children measured only in 1998 and measured in 1998 and 1999, by age group

Age group	Height-for-age Z-score, children measured only in 1998	Height-for-age Z-score, children measured in 1998 and 1999	T statistic on difference in means
6-12 months	-0.24	-0.71	3.20*
12-36 months	-1.48	-1.60	1.26
>36 months	-1.80	-1.83	0.25

Note: \* Means significantly different at the 5 percent level.

Distribution of children in INSP longitudinal sample among treatment-control and eligible-noneligible categories. Table 4 summarizes the distribution of children who are in both rounds of the INSP sample among treatment versus control and eligible for PROGRESA and not eligible categories. The top panel refers to all children, of which there are 663 in households eligible for PROGRESA, fairly evenly split between those listed in treatment versus control households. The bottom panel refers to children ages 12–36 months in August 1998, which is the sample on which this analysis focuses (see below), so it is highlighted in bold. There are 320 such children, divided almost evenly between those listed to be in treatment versus those in control households.

Table 4—INSP longitudinal sample for 1998 and 1999: Numbers of children less than 60 months of age in August 1998 in PROGRESA control and treatment communities who are eligible and not eligible for PROGRESA

	Treatment	Control	Total
Full samp le (children ages 0-6	0 months in August 1998)		
Eligible	337	326	663
row percentage	51%	49%	100%
column percentage	86%	75%	80%
Not eligible	56	109	165
row percentage	34%	66%	100%
column percentage	14%	25%	20%
Total	393	435	828
row percentage	47%	53%	100%
column percentage	100%	100%	100%
Children ages 12-36 months in	August 1998		
Eligible	159	161	320
row percentage	50%	50%	100%
column percentage	90%	78%	84%
Not eligible	17	46	63
row percentage	27%	73%	100%
column percentage	10%	22%	16%
Total	176	207	383
row percentage	46%	54%	100%
column percentage	100%	100%	100%

Representation of PROGRESA treatment. As noted in Section 2, for the preferred representation of having PROGRESA treatment, we use information on whether children have received supplements as well as whether they are in treatment households. The estimates presented in Section 5 indicate that this measure of having received treatment is more consistent with reported child growth than is listed treatment. Three important characteristics of the reports on having received treatment are now noted. First, as will be discussed in Section 5, by this definition, 50–60 percent of children in the 12–36 month age range in eligible households that are listed as receiving PROGRESA treatment (below referred to as "listed treatment") are reported actually to

have "received treatment." Second, 5 percent of the children ages 12–36 months in eligible households in the control localities report having received supplements, so there is a small downward bias in any representation of the PROGRESA treatment that is conditional on being designated as a member of a listed treatment household. Third, the proportion of children in eligible treatment households that have received supplements in fact does not seem to be concentrated in the 4–24 month age range, despite the stated intention to do so (see Section 2). The first column in Table 5 indicates that there is a slightly higher incidence of receiving supplements for children less than 24 months of age, but not a very large difference.

Age range for preferred estimates. The last two columns in Table 5 indicate that the INSP sample is concentrated on the first three to four years of life, perhaps because of the PROGRESA emphasis on this age range. Given this concentration, the stated orientation of the program toward children of less than 24 months of age, that many children less than 12 months are not yet receiving other foods to supplement breastfeeding (and those that are likely to be a selected sample for which breastfeeding is difficult), and that the previous literature suggests that the risk of faltering permanently behind in terms of long-term negative impacts of early childhood malnutrition is thought to continue until about age three (e.g., Martorell 1995, 1999), this investigation concentrates on children 12–36 months of age. (Section 5, however, explores how sensitive these estimates are to changing this age range.)

Table 5—Proportion of children in eligible treatment households who received nutrition supplements and number of children in INSP sample of eligible households, by age groups

	Proportion of children in	Number of children in INSP sample of eligible households		
Age group in months in August 1998	eligible treatment — households who received nutrition supplements	Treatment households	Treatment and control households	
0-12	0.64	108	210	
12-24	0.61	76	151	
24-36	0.52	83	169	
36-48	0.57	69	131	
48-60		1	2	

Basic characteristics of variables used in the analysis. The first column in Table 6 gives the means and standard deviations for all children less than 60 months of age in the INSP sample who are in PROGRESA-eligible households for groups of variables related to child characteristics, parental and household characteristics (further subdivided into mother's characteristics, household head characteristics, household and housing characteristics, resources—per-capita consumption and land owned, whether household received food help, prices, community characteristics, and states of residence. For most of the variables included, there is considerable within-sample variation. As noted in Section 1, the children tend to be stunted much more than wasted. Their parents are characterized by having generally low schooling, undertaking primarily agricultural work, living mostly (70 percent) in formal marriages, and by speaking (in 30 percent of the children considered here) indigenous languages. Households' average over seven

Table 6—Means and standard deviations for children eligible for PROGRESA less than 60 months of age and variables for which the means differ significantly between eligible treatment and control households in August 1998 for all children < 60 months and for those 12-36 months

				T) significantly	
	Mean and	than that for c	ontrol (C) and	d significance of	
	standard			Children betwe	
	deviation for all			months eligible for	
	children < 60	eligible for Pl	eligible for PROGRESA		RESA
	months of age	Listed	Received	Listed	Received
	eligible for	treatment	treatment	treatment	treatment
Variables	PROGRESA	versus control	versus others	versus control	versus others
Child characteristics					
Blood count	11.2 (2.9)				
Height (centimeters)	77.2 (11.2)		C**	C**	C*
Weight (kilograms)	9.9 (2.8)		C**		C*
Body mass index	16.5 (4.1)				
Height-for-age Z-score	-1.5 (1.4)	C**	C*		C*
Weight-for-age Z-score	-1.2 (1.3)	C***	C**		
Weight-for-height Z-score	-0.2 (1.2)				T***
Birth weight (kilograms)	3.1 (0.6)	C***	C***	C**	
Reported birth size	3.0 (1.0)				
Stunted	0.36 (0.48)	T*	$T^*$	T*	$T^*$
Wasted	0.05 (0.22)			C**	
Age (months)	22.2 (13.4)				C***
Sex (female)	0.52 (0.50)				
Parental and household characteristi					
Mother's Characteristics					
Blood count	12.5 (2.6)				
Height (centimeters)	148.2 (5.8)				
Weight (kilograms)	53.3 (10.5)				
Body mass index	24.2 (4.3)				
Age (years)	29.8 (6.6)	T***			
Schooling (grades)	3.6 (2.7)	C*	C*	C**	
Household head characteristics					
Sex male	0.97 (0.17)				
Marital status "open union"	0.27 (0.44)				
Marital status "married"	0.69 (0.46)			C**	
Marital status single	0.005 (0.067)				
Indigenous (speaks dialect or	0.30 (0.46)	C*	C*	C*	C*
indigenous language)	0.20 (0.10)	· ·	C	· ·	C
Agricultural work primarily	0.67 (0.47)				
Nonagricultural worker	0.11 (0.31)				
Self-employed	0.08 (0.26)				
Age (years)	37.8 (11.3)		C*		C**
Literate	0.78 (0.41)		C		C
Schooling (grades)	3.5 (2.6)				
Household and housing characte					
Household size	7.3 (2.5)				
Dirt floors	0.67 (0.47)				T**
Piped water access	0.34 (0.47)	Т*	Т*	Т*	T*
Electricity	0.59 (0.47)	C*	C*	C*	C*
Rooms per household member	0.22 (0.11)	Č	Č	Č	Č
Ln Per capita consumption, land	` '				
Ln per capita consumption	5.01 (0.57)	T*	T*	T*	
En per capita consumption	3.01 (0.37)	1	1	1	

		Whether mean	for treatment (	T) significantly	greater or less
	Mean and	than that for c	control (C) and	d significance of	
	standard			Children between	en 12 and 36
	deviation for all	All children «	< 60 months	months eli	gible for
	children < 60	eligible for Pl	ROGRESA	PROGE	RESA
	months of age	Listed	Received	Listed	Received
	eligible for	treatment	treatment	treatment	treatment
Variables	PROGRESA	versus control	versus others	versus control	versus others
Land owned (area)	2.9 (25.0)	T**			
Whether household received for	od help				
Liconsa (subsidized milk)	0.95 (0.22)	T**			
DIF (school breakfasts, food	0.89 (0.31)	T*	T*	T*	T*
support & community kitchens)	, ,				
Free tortillas	0.995 (0.068)				
Prices	()				
Tomatoes	43.2 (5.4)	C*			
Onions	30.8 (5.9)	C**		C***	
Potatoes	30.9 (7.8)	C		C	
Tortillas	12.5 (2.6)	C*	C*	C*	C*
Rice	30.7 (4.0)	C	C	C	C
Beans	48.6 (6.2)	C*			
Chickens	94.1 (21.1)	C***		C*	
Eggs	45.1 (4.6)	C**		C	
Milk	30.1 (34.7)	C*	C*	C*	
Sugar	25.3 (1.8)	T*	T*	T**	
Oil	47.5 (3.4)	1	1	1	
Community characteristics	17.5 (5.1)				
DICONSA shop	0.20 (0.40)				
DIF health center	0.00 (0.00)				
DIF food	0.45 (0.50)	C*	C*	C*	C*
DIF kitchens	0.10 (0.30)	C	C	C**	· ·
FIDELIST tortilla	0.03 (0.17)	C*	C**	C*	
IMSS clinic	0.07 (0.25)	T*	C	C	
INI shelters	0.002 (0.04)	1			
LICONSA milk	0.15 (0.36)	T*	Т*	T*	Т*
Primary school	0.94 (0.23)	1	1	1	
Sec de Salud clinic	0.33 (0.47)	C**		C**	
Drainage	0.06 (0.23)	C	T**	C	
Electricity	0.72 (0.45)	C*	C*	C**	C*
Piped water	0.72 (0.43)	C.	C.	C	C.
Highway	0.28 (0.45)				
Main road	0.28 (0.43)	Т*	T**	Т*	
Paved road	0.90 (0.30)	C**	1	C*	C**
		C··		C.	C*
Public transportation States	0.41 (0.49)				C*
Puebla	0.26 (0.44)	Т*		Т*	
Oueretaro	0.28 (0.44)	1 '		T*	
Queretaro San Luis				1 "	
	0.17 (0.37)	C*	C*	C*	C*
Veracruz	0.16 (0.37)	C*	C*	U"	C*

<sup>&</sup>lt;sup>a</sup> The number of observations varies among the variables because of missing observations. The maximum number of children who are eligible for PROGRESA in the sample is 663 for all children under 60 months and 320 for children between 12 and 36 months.

<sup>&</sup>quot;T" indicates that mean for treatment >mean for control (and vice versa for "C"). Significance levels are \* = 5 percent, \*\*= 10 percent and \*\*\* = 15 percent for two-sample t-tests with equal variances. A shaded box indicates no significant difference in means at the 15 percent or lower confidence level.

23

members who tend to live in fairly crowded houses, the majority of which have dirt floors and no access to piped water. The communities have varied social services, infrastructure, and transportation links; while almost all having primary schools, the majority do not have drainage or health clinics.

Assignment to treatment versus control samples. If households were assigned randomly to treatment versus control samples with regard to all relevant observed and unobserved determinants of child health/height, simple comparisons of child health/height would reveal the treatment impacts. PROGRESA attempted to assign communities randomly to treatment versus control communities (localities) in its basic evaluation samples. Based on the initial rounds of data collected, Behrman and Todd (1999) conclude that assignment was basically random at the community level, but that at the household level there are more rejections of random assignment than would be expected by chance. As discussed above, the INSP longitudinal data that are central for the present study are a subsample of the larger PROGRESA evaluation sample, so it is not clear to what extent these results carry over to the INSP data.

The last four columns in Table 6 summarize whether means for all the variables in the INSP sample differ significantly between the listed treatment and control children and between the received treatment and others, separately for all children less than 60 months and for children 12–36 months of age in August 1998. For many of the variables, there are not significant differences. But there are significant differences more often than would occur by chance, suggesting that assignment was not random. Moreover, in the fairly high proportion of cases in which there are significant differences for child

nutrition indicators, the children in the control households almost always were initially better off than those in the treatment households. This means that a simple comparison between control and treatment households in the second round would tend to *understate* the impact of PROGRESA. For this reason it would appear that such simple cross-sectional comparisons are not always reliable. By taking the approach outlined at the end of Section 2 for the preferred estimates, this analysis will control for the basic observed and unobserved differences between the treatment and control children, including that the control sample apparently tended to start with somewhat better health and nutritional status than did the treatment sample.

#### 5. ESTIMATES OF PROGRESA IMPACT

#### ESTIMATED IMPACT ON CHILD HEIGHT

#### *OLS Cross-Sectional Estimates*

If the individual children in the sample were selected randomly for the treatment versus control group, then a simple comparison of their heights in 1999 would reveal the impact of the PROGRESA treatment. The exploration in Section 4 suggests that assignment of children between treatment and control probably was not random. Nevertheless, it is instructive to see what cross-sectional comparisons suggest for children who received treatment versus those who did not. Table 7 presents three OLS cross-sectional estimates for 1999 for children 12–36 months of age in August 1998 in

Table 7—OLS estimates of impact of treatment on height in 1999 for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA

Received treatment	-1.82 (2.3)*	-1.10 (1.7)**	-0.85 (1.2)
Ln child age		18.2 (13.1)*	19.1 (14.3)*
Constant	88.7 (228.9)*	23.4 (4.7)*	-32.8 (0.9)
Joint test for all child coefficients			F(4, 250) = 54.35 $Prob > F = 0.0000$
Joint test for all household and parental coefficients			F(23, 250) = 1.90 $Prob > F = 0.0093$
Joint test for all price coefficients			F(11, 250) = 0.96 $Prob > F = 0.4799$
Joint test for all community coefficients			F(12, 250) = 1.42 Prob > F = 0.1581
Joint test for all state coefficients			F(5, 250) = 1.59 Prob > F = 0.1642
Joint test for all coefficients	F(1, 318) = 5.41 $Prob > F = 0.0207$	F(2, 317) = 90.10 Prob > F = 0.0000	F(69, 250) = 5.88 Prob > F = 0.0000
Number of observations	320	320	320
R squared (adjusted)	0.0136	0.3584	0.3774
Root Mean Squared Error	6.03	4.86	4.79

Notes: Absolute t values in parentheses to right of point estimates. \* indicates significance at 5 percent level, \*\* at 10 percent level, and \*\*\* at 15 percent level. For the estimates in the right column dummy variables are used for missing prices or ln consumption per household member (which effectively controls for the means for the missing values for these variables). Prices are median locality prices of tomatoes, onions, potatoes, tortillas, rice, beans, chicken, milk, vegetable oil, eggs and sugar.

households eligible for PROGRESA of relation (1B).<sup>8</sup> The first includes only whether the children received treatment. The second adds the logarithm of child age. The third

<sup>8</sup> Below we explore the robustness of our fixed-effects estimates to changes in the ages of children considered, to alternative representation of the PROGRESA treatment, and to alternative treatment of missing observations. Here we report on the OLS estimates using the choices that we prefer for each.

26

includes the childbirth weight and sex and all of the household and parental background, community, and state variables that are discussed in Section 4 (Table 4).

What do these estimates suggest about the impact of PROGRESA on child growth? At a first glance, the answer is not encouraging. For all three estimates, ranging from no to a large number of controls, the estimated PROGRESA treatment effects are negative, though significantly nonzero at the 5 percent level only in the first estimate and at the 10 percent level in the second. If indeed the children are randomly assigned to treatment rather than to the control group, these estimates suggest that, if anything, PROGRESA had a negative effect on child growth.

The pattern across the estimates indicates, however, that when there are additional controls the estimates become smaller in absolute magnitude and less precise, though the estimate is still negative in the last column that includes a large number of controls. The relatively large and significant negative coefficient estimate in the first column, thus, may reflect that treatment is positively correlated with characteristics that are associated with less child growth, so the coefficient estimate for treatment is biased. Among the controls included in the estimates in the third column, only the groups for the child and for parental and household characteristics are statistically significant at the standard 5 percent level, not prices or community or state characteristics. The particular child and parental and household characteristics that have significant coefficient estimates at least at the 10 percent level, despite the substantial multicollinearity that results from including so many right side variables, are positive ones for child age, child being a boy, child birth weight, mother's height, rooms per household member, and not having a dirt floor. A

logit estimate of receiving treatment on these characteristics yields a relation that is significant at the 5 percent level and that has negative coefficient estimates on all these characteristics.

#### Basic Individual Child Fixed Effects Estimates

The examination of the data in Section 4 suggests, however, that children may not have been assigned randomly to treatment versus control groups in terms of observed characteristics, importantly including their initial health and nutrition status. This also suggests that they might not have been so assigned in terms of characteristics that are not observed in the data. If either of these possibilities holds, then cross-sectional estimates of the treatment effects, such as these, are biased.

Table 8 gives two individual child-fixed effects estimates, akin to relation (1C), using the panel data on individual children in 1998 and 1999. The estimates in the first column use a simple specification that controls for the observed nonlinear child age effect and a possible secular trend, and adjusts for differing lengths of time between measurements. The estimates in the second column also control for fluctuations in household consumption expenditure per household member and in food prices, as discussed in Section 3 above. It should be remembered that these coefficients capture the

\_

 $<sup>^{9}</sup>$  Chi<sup>2</sup>(8)=15.73, prob>chi<sup>2</sup> = 0.0464. Coefficient estimates (absolute z statistics) are -0.29 (1.0) for the child being a boy, -1.27 (1.9) for ln child age in months, -0.55 (1.8) for child birth weight in kg, -0.26 (1.8) for relative birth size, -1.24 (1.0) for rooms per household member, and -0.57 (1.8) for not having a dirt floor.

Table 8—Individual child fixed effects estimates of impact of treatment on height for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA

Received Treatment	0.95 (2.4)*	1.02 (2.6)*
Ln child age	6.3 (4.1)*	6.8 (4.4)*
Ln consumption per household member		0.37 (1.2)
Trend	5.3 (7.9)*	5.49 (6.1)*
Mean fixed effect	60.2 (12.4)*	55.0 (8.2)*
Joint test for price coefficients		F(11, 299) = 1.28 $Prob > F = 0.2340$
Joint test for all coefficients	F(1, 317) = 804.41 $Prob > F = 0.0000$	F(21, 299) = 117.60 $Prob > F = 0.0000$
Number of observations	640	640
R squared: within between overall	0.8839 0.4276 0.5056	0.8920 0.3377 0.5005
sigma_u sigma_e rho (fraction of variance due to u_i)	5.01 2.12 0.85	5.04 2.10 0.85
F test that all u_i=0:	F(319, 317) = 8.53 $Prob > F = 0.0000$	F(319, 299) = 8.18 $Prob > F = 0.0000$
Hausman specification test	chi2(3)=49.04 Prob>chi2 = 0.0000	chi2(21)=136.89 Prob>chi2 = 0.0000

Notes: Absolute t values in parentheses to right of point estimates. \* indicates significance at 5 percent level, \*\* at 10 percent level, and \*\*\* at 15 percent level. For the estimates in the right column dummy variables are used for missing prices or ln consumption per household member (which effectively controls at the means for the missing values for these variables). Prices are median locality prices of tomatoes, onions, potatoes, tortillas, rice, beans, chicken, milk, vegetable oil, eggs and sugar.

impact of possible transitory changes in these variables. This is not necessarily equivalent to a change in the value of their "permanent" or long-term level. Further, to the extent to which these regressors are measured with error, their coefficients will be biased downward toward zero. Both estimates control for unobserved fixed child (or child's

family, household or community) characteristics. The F-tests in the penultimate row suggest that for all of these specifications the probability is very low (less than 0.0000) that these estimates would be obtained were all the individual fixed effects zero. The Hausman specification test results in the last row indicate that the probabilities also are very low that the unobserved effects are random rather than correlated with the right-side variables. The individual fixed effects account for about 85 percent of the variance in the disturbance terms and for the majority of the explained variance (which overall is above half).

The striking result is that in these estimates receiving treatment has a significant *positive* estimated impact on child growth—in sharp contrast to the OLS cross-sectional estimates in Table 7. Conditional on the individual unobserved child-fixed effects that prevail in the estimates in Table 8, the OLS cross-sectional estimates in Table 7 are biased downward to the point of being negative rather than positive by negative correlations between receiving treatment and unobserved determinants of child growth.

The estimates in Table 8 indicate that those children receiving treatment experienced growth per year of about one cm greater than those who did not. This is about a sixth of the mean growth per year and about a third of the standard deviation in that growth per year that would have been experienced by those in this sample in the absence of PROGRESA.<sup>10</sup> These estimates, thus suggest that PROGRESA had an

\_

<sup>&</sup>lt;sup>10</sup> This calculation is based on subtracting an average of 1 cm from the annual growth for every child in the sample who received treatment and then calculating the summary statistics for annual growth for the sample: mean of 7.9 centimeters, median of 7.4 centimeters, standard deviation of 3.1 centimeters, and range from 0.3 to 21.3 centimeters.

important impact on the children who received treatment in the critical 12–36 month age range. The estimated effects are about the same whether there are controls for transitory fluctuations in the log of consumption expenditure per household member and in prices—which as a group do not have effects that are significantly nonzero.

As noted in Section 3, the fixed effects estimates control for all fixed effects, whether they are observed in the data or not. An interesting question is the extent to which the fixed effects implied by the estimates in Table 8 are related to observed characteristics of children in the sample, their parents and households in which they live and the prices and other characteristics of the communities and states in which they live.

Table 9 provides some insight into this question. The dependent variable in this regression is the estimated child-specific fixed effect that was obtained when estimating the results reported in Table 8. Because there are so many possible correlates of these fixed effects, Table 9 only reports significance tests for groups of variables and for coefficient estimates for individual variables for which t-tests indicate significance at least at the 15 percent level (as well as the names of such variables for which there is not such significance). These indicate that the child specific fixed effect is associated at least at the 1 percent level with each of the major groups of variables that are included—child characteristics, household and parental characteristics, prices, and community and state characteristics. Despite the presence of substantial multicollinearity among these regressors (due to the many characteristics included in this regression), a number of the individual coefficient estimates are significantly nonzero at least the 15 percent level.

Table 9—Relation between estimated individual child growth fixed effects from Table 8 and observed fixed child, household and parental, price, community and state characteristics for children 12-36 months in August 1998 and eligible for PROGRESA

Child characteristics F(4, 251) = 759.94, Pro	b > E = 0.0000		
Birth weight (kilograms)	$00 > \Gamma = 0.0000$	ln age (months)	6.6 (54.7)*
Reported birth size		Sex (female)	0.0 (34.7)
Parental and household characteristics F(23	251) - 2.64 Pro		
Mother's characteristics	, 231) – 2.04, F10	$00 > \Gamma = 0.0001$	
Height (centimeters)		Aga (vants)	
Weight (kilograms)		Age (years) Schooling (grades)	
Body mass index		Schooling (grades)	
Household head characteristics Sex male		Nonagricultural worker	
	-0.29 (1.5)***	•	
Marital status "open union"	-0.29 (1.3)***	Self-employed	0069 (2.4)*
Marital status "married"		Age (years)	0068 (2.4)*
Marital status single		Literate	
Indigenous (speaks dialect or indigenous language)		Schooling (grades)	
Agricultural work primarily			
Household and housing characteristic	s		
Household size		Electricity	-0.17 (2.6)*
Dirt floors		Piped water access	0.16 (2.4)*
Rooms per household member		r	
Ln Per capita consumption, land owner	ed		
In per capita consumption	0.31 (4.7)*	Land owned (area)	
Prices $F(11, 251) = 22.37$ , $Prob > F = 0.0000$	, ,	,	
Tomatoes	0.027 (3.8)*	Chickens	
Onions	0.053 (7.0)*	Eggs	0.025 (2.9)*
Potatoes	-0.023 (3.1)*	Milk	-0.005 (7.4)*
Tortillas	-0.019 (6.6)*	Sugar	
Rice	, ,	Oil	
Beans	0.070 (9.2)*		
Community characteristics $F(12, 251) = 3.19$	P,  Prob  > F = 0.00	03	
DICONSA shop		Sec de Salud clinic	
DIF health center		Drainage	0.35 (1.9)**
DIF food	-0.28 (3.9)*	Electricity	
DIF kitchens		Piped water	-0.20 (2.6)*
FIDELIST tortilla		Highway	0.11(1.6)***
IMSS clinic	-0.27 (1.9)**	Main road	
INI shelters		Paved road	
LICONSA milk		Public transportation	
Primary school		-	
States $F(5, 251) = 3.37$ , $Prob > F = 0.0058$			
Guerrero	-0.26 (1.4)***	Queretaro	
Hidalgo		San Luis	
Puebla		Veracruz	-0.35 (3.4)*
Number of observations = 320		Adjusted R-squared = $0.9408$	
F(68, 251) = 75.49, Probability $> F = 0.0000$		Root Mean Squared Error = 0.385	
Notes: Dependent variable is fixed effect fr	om estimates in r	ight hand column in Table 9. In addition	n to the veriables

Notes: Dependent variable is fixed effect from estimates in right-hand column in Table 8. In addition to the variables included in this table there also are variables to control for the mean of missing observations for some variables. Only coefficient estimates that are significantly nonzero at at least the 15 percent level are included. Absolute values of t-tests are in parentheses to the right of the point estimates. Significance levels are \* = 5 percent, \*\*= 10 percent and \*\*\* = 15 percent for two-sample t-tests with equal variances.

Positive associations are reported for the log of child's age, household access to piped water, log consumption expenditures per household member, some prices (for tomatoes, onions, beans, and eggs), and drainage and a highway at the community level. Negative coefficients are reported for the household head being in an "open" martial union rather than married, for older household heads, for the household having electricity, for some prices (for potatoes, tortillas, and milk), for some community characteristics (having a DIF food program, an IMSS clinic, and piped water), and for being in Guerrero or Veracruz rather than in the other four states included in the sample. Despite the difficulties in interpreting such reduced-form relations (e.g., the prices responses may include effects of being producers as well as consumers of these items), these patterns generally seem to be plausible. 11 All in all, the observed characteristics included in Table 9 are consistent with a considerable portion of the variance in the individual child-fixed effects, with an adjusted  $R^2$  of 0.94. The residual in this estimate, while strongly positively correlated with receiving PROGRESA treatment, is negatively correlated with child height—so the unobserved part of the child fixed effects may be underlying the sign reversal between the OLS cross-sectional estimates in Table 7 and the fixed effects estimates in Table 8 (even though the observed characteristics are associated with most of the variance in the fixed effects).

<sup>&</sup>lt;sup>11</sup> Two possible exceptions are the negative coefficient estimate for the household having electricity and the opposite signs for the household versus the community having piped water.

Some Aspects of the Robustness of the Fixed Effects Estimates

We consider questions relating to three aspects of our estimates—missing observations for right-side variables, the age range of children, and the empirical representation of the PROGRESA treatment.

One question is how to address missing observations on individual variables, particularly on prices. In alternative estimates, we have explored the option of simply dropping from the estimates all of the observations in which the value for any included variable is missing. If we do so, the estimated coefficient on the PROGRESA treatment increases by almost 60 percent in the estimates in the right-hand column of Table 8. But this apparently reflects sample selection because there is a similar increase of almost 40 percent in the estimated PROGRESA treatment effect for the estimates in the left-hand side relation in Table 8 if it is limited to observations with complete information on prices. And there is a fairly substantial reduction in observations from 640 (320 children with information on both rounds) to 529. Therefore, we adopt the alternative of not dropping any observations because of missing data on individual variables, but controlling for the mean of such missing data through dummy variables.

A second question concerns the age range of children who are included in our basic estimates. As we discuss in Section 2 above, there are *a priori* reasons based on the age of weaning and on the nature of the PROGRESA program to focus on roughly the 12–36 month age range. But it is of interest to know what happens for other age ranges. Table 10 summarizes some important aspects of alternative estimates for which the

specification is identical to that in the right-hand side of Table 8, but the sample is limited to different age ranges for the children.

The first three columns in Table 10 give estimates for 0–12, 12–36 and 36–60 months, respectively. The column for the 12–36 month age range, on which this paper focuses, is in bold for easy reference. These three columns suggest that the impact of PROGRESA indeed is centered on the 12–36 age range; only for this age range, among

Table 10—Individual child fixed effects estimates of impact of treatment on height for children in August 1998 who are in households eligible for PROGRESA for alternative age ranges<sup>a</sup>

	Age range (months)							
	0-12	12-36	36-60	6-36	12-30	12-24	12-42	12-48
Received treatment	0.26 (0.6)	1.02 (2.6)*	-0.36 (0.7)	0.84 (2.5)*	0.98 (2.2)*	0.78 (1.3)	0.70 (2.1)*	0.62 (2.0)*
Ln child age	4.1 (13.6)*	6.8 (4.4)*	-7.4 (0.5)	4.3 (6.5)*	7.2 (3.6)*	6.8 (2.1)*	7.3 (5.8)*	6.9 (6.0)*
Joint test for all coefficients (Prob $> F = 0.0000$ for all cases)	F(22, 188)= 182.16	F(21, 299)= 117.60	F(20, 113)= 51.96	F(22, 418)= 172.57	F(21, 204)=9 6.52	F(21, 130)= 59.74	F(21, 376)= 149.84	F(21, 430)= 165.70
Number of observations	420	640	266	880	450	302	794	902
R squared: within between overall	0.9952 0.6143 0.7566	0.8920 0.3377 0.5005	0.9019 0.0370 0.3038	0.9008 0.5987 0.5250	0.9086 0.2491 0.5315	0.9061 0.1921 0.5160	0.8933 0.4990 0.5295	0.8900 0.5587 0.5299
sigma_u sigma_e rho <sup>a</sup>	3.75 1.91 0.29	5.04 2.10 0.85	4.87 1.83 0.88	5.79 2.10 0.88	4.42 2.02 0.83	4.54 2.19 0.81	5.38 2.02 0.88	5.68 2.02 0.89
F test that all $u_i=0$ (Probability > F = 0.0000 for all columns)	F(209, 188) = 5.19	F(319, 299) = 8.18	F(132, 113)= 9.92	F(439, 418)= 7.48	F(224, 204)= 7.18	F(150, 130)= 5.55	F(396, 376)= 9.33	F(450, 430)= 9.22

Notes: The specifications for all columns are identical to that in the right-side column in Table 8 (which is reproduced here in bold) but the age range for children in the sample varies as indicated. Absolute t values in parentheses to right of point estimates. \* indicates significance at 5 percent level, \*\* at 10 percent level, and \*\*\* at 15 percent level.

<sup>&</sup>lt;sup>a</sup> rho is the fraction of variance due to u\_i.

the three considered, is the estimated effect significantly nonzero. The last five columns in Table 10 give estimates for the 12–36 month age range with one or the other limit of this age range changed by six or 12 months. These estimates tend to suggest somewhat smaller estimated impacts of PROGRESA, with point estimates from 0.62 to 0.98 as compared with 1.02 for the 12–36 month range. For all the different age ranges considered in this table, the overall relations and the child fixed effects in particular are significantly nonzero at high levels (with probabilities > than 0.0000).

A third question concerns the representation of the PROGRESA treatment variable. Tables 11a and 11b consider four representations of the PROGRESA treatment variable, all for the same 12–36 month age range:

- (1) Received treatment: defined by being eligible for treatment and being in a treatment community and whether the household respondent indicated that the child had received benefits in at least one month. This is the preferred representation used in the tables above.
- (2) Listed treatment: defined as residing in a household eligible for PROGRESA and being located in a treatment community.
- (3) Received treatment adjusted for intensity (1): defined by the product of being eligible for treatment, being in a treatment community, and the number of months the child is reported in the 1999 survey as having received treatment (normalized to range from 0 to 1.0).

(4) Received treatment adjusted for intensity (2): defined by the product of being eligible for treatment, being in a treatment community, and the number of months the child is reported in the 1999 survey as having received treatment up to a maximum of 12 months (some children are reported to have received the supplement for more than 12 months. This too is normalized to range from 0 to 1.0.

Table 11a gives the means and standard deviations for each of these four representations and the correlations among them. Only about quarter-to-a-third of the sample (0.24 to 0.31) received treatment in the sense defined by the first representation, which is about 50–60 percent of the number indicated for "listed treatment." The three "received treatment" measures are fairly highly correlated (with r at least equal to 0.9), but are much more weakly correlated with the "listed treatment" measure r between 0.53 and 0.58). Therefore, "listed treatment" may be only a moderately useful proxy for having received treatment. Table 11b gives estimates of the specification in the rightmost column in Table 8 for the same age range of 12–36 months, but with the four alternative representations of PROGRESA treatment. In most respects the summary statistics for the four estimates look very similar, with high levels of significance for all the coefficient estimates and for the fixed effects and quite similar R's squared and decompositions of the disturbances between the fixed effects and the random term. The one important difference, however, is that the "listed treatment" representation is much

Table 11a—Means, standard deviations, and correlations for four alternative representations of PROGRESA treatment for children in eligible households who were 12-36 months of age in August 1998

			Correlations			
			1	2	3	4
	Mean	Standard Deviation	Received treatment	Listed treatment	Received treatment adjusted for intensity (1)	Received treatment adjusted for intensity (2)
1. Received treatment	0.24-0.31 <sup>a</sup>	0.43	1.00	0.58	0.90	0.96
2. Listed treatment	0.50	0.50		1.00	0.53	0.57
3. Received treatment adjusted for intensity (1) <sup>b</sup>	0.10	0.19			1.00	0.95
4. Received treatment adjusted for intensity (2) <sup>b</sup>	0.23	0.40				1.00

<sup>&</sup>lt;sup>a</sup> This range reflects the range of possibly assumptions for the treatment of missing observations for the question regarding months of usage (of which there are 21 among the 320 children used for our basic analysis). Whether we treat these individuals as missing or as receiving no benefits changes the coefficient estimate for received treatment in Table 8 only in the third digit (by less than .03).

more imprecisely estimated than the "received treatment" representations and, in the case in which comparisons of the coefficient estimates make sense, is less that half the magnitude of the "received treatment" representation (i.e., option 1 versus 2). It would appear that the "listed treatment" variable is a noisy representation of actually receiving treatment with the result that the use of "listed treatment" is likely to lead to a bias toward zero of the true effect. Put another way, the distinction between being "listed for treatment" and "actually receiving treatment" is critical in assessing PROGRESA's impact.

<sup>&</sup>lt;sup>b</sup> These are normalized so that their ranges go from 0 to 1, so their means and standard deviations are not comparable to those for the first two representations.

Table 11b—Individual child fixed effects estimates of impact of treatment on height for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA for alternative representations of PROGRESA treatment<sup>a</sup>

	Alternative representations of PROGRESA treatment				
	1. Received treatment	2. Listed treatment	3. Received treatment adjusted for intensity (1) <sup>b</sup>	4. Received treatment adjusted for intensity (2) <sup>b</sup>	
Treatment coefficient	1.02 (2.6)*	0.46 (1.3)	0.24 (2.0)*	1.00 (2.2)*	
Joint F(21, 299) test for all coefficients	117.60  Prob > F = 0.0000	115.49 Prob > F = 0.0000	103.96 Prob > F = 0.0000	104.39 Prob > F = 0.0000	
R squared: within between overall	0.8920 0.3377 0.5005	0.8903 0.3802 0.5112	0.8870 0.3535 0.5015	0.8875 0.3443 0.4977	
sigma_u sigma_e rho <sup>c</sup>	5.04 2.10 0.85	4.98 2.12 0.85	5.01 2.16 0.85	5.03 2.16 0.84	
F(319, 299) test that all u_i=0:	8.18 Prob > F = 0.0000	8.07 Prob > F = 0.0000	7.90 Prob > F = 0.0000	7.91 Prob > F = 0.0000	

<sup>&</sup>lt;sup>a</sup> Specification and sample is same as for right-most column in Table 8 and the first column here, highlighted in bold, is the same as that column. Absolute t values in parentheses to right of point estimates. \* indicates significance at 5 percent level, \*\* at 10 percent level, and \*\*\* at 15 percent level. For the estimates in the right column dummy variables are used for missing prices or ln consumption per household member (which effectively controls at the means for these variables).

c rho is the fraction of variance due to u\_i.

#### Interactions Between PROGRESA Treatment and Observed Characteristics

The estimates discussed to this point assume that, controlling for unobserved fixed effects, transitory fluctuations in consumption expenditure per household member and in prices, and child's age, the impact of PROGRESA is the same on all children who were in the 12–36 age range in August 1998. But the conceptual framework in Section 3

These are normalized so that their ranges go from 0 to 1, so their means and thus their coefficient estimates are not directly comparable to those for the first two representations.

suggests that the impact may vary depending on characteristics of the child, his or her family, and the community in which he or she lives.

Accordingly, we present estimates that allow the impacts of receiving the PROGRESA treatment to vary by child, parental and household, and community and state characteristics. We do so by including in the specification interactions between such characteristics and having received treatment, in addition to the direct effect of having received treatment itself. We first have explored such interactions, one variable at a time, for most of the variables that are included in Table 4.<sup>12</sup> The subset of these estimates in which the interaction has a coefficient estimate that is significantly nonzero at least at the 15 percent level is included in Panel A of Table 12. We then explore what subset of these interactions remains significantly nonzero when they are included in combination, as summarized in Panel B of Table 12. These latter estimates are more interesting because they are more robust to the inclusion of other controls.

The significant estimates in Panel B of Table 12 suggest that three household characteristics and two community characteristics affect the magnitude of the PROGRESA treatment differentially. The positive estimates for two of the household characteristics, speaking an indigenous language and household size, may reflect that poorer households that are more likely to speak an indigenous language and to be larger are more constrained by market imperfections. Therefore, their children benefit more

<sup>&</sup>lt;sup>12</sup> We have not included the child characteristics that refer to anthropometric status subsequent to birth because these would seem clearly to reflect behavioral choices of the types under investigation, nor do we include prices.

Table 12—Summary of interaction effects between receiving treatment and child, household, community and state characteristics<sup>a</sup>

			Panel B. Adding all interactions that		
Panel A. Adding one interaction at a time			remain significant in combination		
	Coefficient			·	
	estimate for	Coefficient			
Variable interacted with	received	estimate for	Variable interacted with	Coefficient	
received treatment	treatment	interaction	received treatment	estimate	
None (same as in Table 8)	1.02 (2.6)*		Constant (i.e., simply received treatment)	-2.89 (2.0)*	
Child birth weight	0.27 (0.4)	0.37 (1.5)***	Child birth weight		
Mother's schooling	0.41 (0.7)	0.19 (1.4)***	Mother's schooling		
Household head speaks indigenous language	0.61 (1.4)	1.99 (2.3)*	Household head speaks indigenous language	2.29 (2.8)*	
Household head primarily agricultural worker	-0.08 (0.1)	1.76 (2.4)*	Household head primarily agricultural worker		
Household head primarily self-employed	1.17 (2.9)*	-2.37 (1.7)**	Household head primarily self-employed		
Household head's schooling	-0.49 (0.8)	0.47 (3.3)*	Household head's schooling	0.45 (3.1)*	
Household size	-0.61 (0.5)	0.24 (1.5)***	Household size	0.35 (2.3)*	
Community has DIF food program	0.66 (1.6)***	2.27 (2.4)*	Community has DIF food program	3.05 (3.1)*	
Community has piped water	1.73 (3.6)*	-1.89(2.6)*	Community has piped water	-2.39 (2.0)*	
Community has highway	1.42 (3.2)*	-1.68 (2.1)*	Community has highway		
Hidalgo state	1.51 (3.2)*	-1.52 (2.0)*	Hidalgo state		
Queretaro state	0.76 (1.9)**	3.19 (2.4)*	Queretaro state		

The specification or each estimate is identical to that in the right-most column of Table 8 for the same 12-36 month age group (for which the estimate is presented in the first row of this table) except that, one at a time, interactions with "received treatment" are added for child age, sex, and birth weight and for all the parental and household, community and state variables that are listed in Table 4 above in Panel A (and a group of such interactions are added in Panel B). Only the interactions that are statistically significant at least at the 15 percent level are presented here. Absolute t values in parentheses to right of point estimates. \* indicates significance at 5 percent level, \*\* at 10 percent level, and \*\*\* at 15 percent level.

from identical options presented by PROGRESA than do children from initially somewhat better off households. The positive estimate for household head's schooling, on the other hand, tends to suggest that somewhat better-off households have greater gains, perhaps because they are better able to process the necessary information to benefit more from PROGRESA. It is interesting to note that it is the household head's schooling

that is significant and not the schooling of the children's mothers even though 97 percent of the household heads in this sample are male. This contrasts with widely-held perceptions that mothers' schooling is particularly important for child development (e.g., Strauss and Thomas 1995).

With regard to the community characteristics, impacts are larger in communities without access to piped water. This is consistent with PROGRESA, having greater effects not only among poorer households but in communities that are poorer with regard to the general health and sanitation environment. The other community variable that had a significant interaction—whether the community has a DIF food program—also may be representing poorer communities if indeed poorer communities are more likely to have such programs. There is some evidence that indeed these programs do tend to be placed in poorer and more malnourished communities. Mean household per-capita expenditure and per-capita caloric consumption both are lower in communities with a DIF food program than in those without such a program.<sup>13</sup>

Thus these estimates suggest that, among children who receive PROGRESA treatment, the effects tend to be larger for poorer households in poorer communities but

<sup>&</sup>lt;sup>13</sup> For the sample on which we focus, the means for children's households from communities with versus without DIF food programs are 158 versus 173 pesos for consumption expenditure per household member (significantly different at the 15 percent level) and 1,789 versus 2057 calories per household member (significantly different at the 5 percent level).

that children in households in which household heads have more schooling also tend to benefit more.<sup>14</sup>

### ESTIMATED IMPACT ON THE PROBABILITY OF STUNTING

The analysis in most of the previous section pertains to the average impact of PROGRESA on child growth, thought the last part of the section investigates possible differences among children in treatment effects. But of particular interest is what are the effects on the children who are faring most poorly—those who are more than two standard deviations below the norms or are stunted. Among the children age 12–36 months in August 1998 who were from households eligible for PROGRESA, 44 percent were stunted. A year later, 41 percent were stunted, including 76 percent of those who were stunted in 1998.

Such data suggest that PROGRESA may have had some small impact on stunting, but it is desirable to go further than such a summary by investigating the probability of being stunted parallel to the investigation of the determination of child growth in the previous section. The available data limit the extent to which such an exploration is possible. In particular, for the fixed effects logit the only observations that affect the estimates are those in which children change from being stunted to nonstunted or from nonstunted to stunted between the two rounds. The numbers who did so are very small.

<sup>&</sup>lt;sup>14</sup> These are suggestive only because there is no significant association with many indicators of household income and wealth such as household consumption expenditure per household member, housing characteristics, etc. (and similarly with regard to many community indicators).

Among the children age 12–36 months in August 1998 who were from households eligible for PROGRESA, 24 percent of those stunted in 1998 were not stunted in 1999 (34 children) and 14 percent who were nonstunted in 1998 were stunted in 1999 (25 children).

Tables 13 and 14 are parallel to Tables 7 and 8, but present logits for stunting instead of estimates for child height. The parallel holds not only for the organization of the tables, but also for the implications of the estimates. The cross-sectional logit estimates for 1999 in Table 13 provide no support for the proposition that PROGRESA reduced child stunting. In fact, all the coefficient estimates of the received treatment variable are the wrong sign (though not significantly different from zero). The fixed effects logit estimates in Table 14, in sharp contrast, indicate significant negative effects of PROGRESA treatment on child stunting. Thus, though qualifications are necessary because of the small number of observations noted above (which may be related to the fairly different values of the point estimates with changes in specification in Table 14), these explorations suggest that (1) the cross-sectional results are basically misleading because there are important unobserved fixed effects and (2) once there is control for the fixed effects, PROGRESA treatment appears to have had a significant effect on reducing child stunting as well as on increasing child growth on the average.

Table 13—Logit estimates of impact of treatment on stunting in 1999 for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA<sup>a</sup>

Received treatment	0.28 (1.1)	0.33 (1.3)	0.14 (0.3)
Ln child age		1.19 (2.0)*	1.00 (1.2)
Constant	-0.44 (3.3)*	-4.71 (2.2)*	15.6 (0.7)
Joint test for all child coefficients			chi2(4) = 14.76 Prob > F = 0.0052
Joint test for all household and parental coefficients			chi2(22) = 42.80 Prob > F = 0.0050
Joint test for all price coefficients			chi2(11) = 21.93 Prob > F = 0.0249
Joint test for all community coefficients			chi2(12) = 13.89 Prob > F = 0.3075
Joint test for all state coefficients			chi2(5) = 7.99 Prob > F = 0.1569
Joint test for all coefficients	LR chi2(1) = $1.15$ Prob > F = $0.0027$	LR $chi2(2) = 5.24$ Prob > F = 0.0727	LR $chi2(65) = 151.48$ Prob > F = 0.0000
Number of observations	320	320	314
Psuedo R squared	0.0027	0.0121	0.3581

<sup>&</sup>lt;sup>a</sup> Absolute z values in parentheses to right of point estimates. \* indicates significance at 5 percent level, \*\* at 10 percent level, and \*\*\* at 15 percent level. Prices are median locality prices of tomatoes, onions, potatoes, tortillas, rice, beans, chicken, milk, vegetable oil, eggs and sugar. For the estimates in the right column dummy variables are used for missing prices or ln consumption per household member (which effectively controls at the means for the missing values for these variables).

Table 14—Individual child fixed effects Logit estimates of impact of treatment on stunting for children 12-36 months of age in August 1998 who are in households eligible for PROGRESA<sup>a</sup>

Received Treatment	-1.50 (2.5)*	-3.54 (2.6)*
Ln child age	-0.02 (0.0)	1.9 (0.5)
Ln consumption per household member		-0.02 (0.0)
Trend	0.23 (0.2)	1.60 (0.6)
Joint test for price coefficients		chi2(11) = 6.37
		Prob > F = 0.8476
Joint test for all coefficients	LR $chi2(3) = 8.25$	LR $chi2(21) = 33.48$
	Prob > F = 0.0412	Prob > F = 0.0411
Number of observations	118	118

<sup>&</sup>lt;sup>a</sup> Absolute t values in parentheses to right of point estimates. \* indicates significance at 5 percent level, \*\* at 10 percent level, and \*\*\* at 15 percent level. Prices are median locality prices of tomatoes, onions, potatoes, tortillas, rice, beans, chicken, milk, vegetable oil, eggs and sugar. For the estimates in the right column dummy variables are used for missing prices or ln consumption per household member (which effectively controls for the means for the missing values for these variables).

### 6. LONGER-TERM IMPACT

This study has focused only on estimating the impact of PROGRESA on child growth. But this impact is of interest, as noted in the introduction, because it may relate to longer-term health and nutritional status and productivity. There are at least four channels through which the component of the PROGRESA program that affects child health/height can affect lifetime earnings: (1) by increasing cognitive skills as an adult (conditional on years of schooling completed) that directly affect earnings, (2) by increasing physical stature as an adult that directly affects earnings, (3) by increasing the years of completed schooling that directly affect earnings and the age of school completion, and (4) by changing the age of school completion without changing the grades of schooling completed. For the program to have impact through channels (3) and (4), we are assuming that improved cognitive skills and nutrition as a child facilitates earlier entry into school, lower repetition rates, and more years of schooling completed.

There is piecemeal empirical evidence of significant effects through all four of these channels for other developing countries:

<sup>&</sup>lt;sup>15</sup> Reductions in the age at which a given grade of school is completed increase the benefits because they permit obtaining post-schooling benefits sooner and longer. Such reductions may occur because of entry into school when younger and/or because of higher progression rates through grades while in school. These may be quite important. For example, in another developing country context, Behrman and Knowles (1999) find that the strongest association between parental family income and schooling success is with the progression rate through school, neither with the grades completed nor the cognitive achievement conditional on grades completed.

- (1) on the impact of adult cognitive achievement on wages: Alderman et al. (1996) for rural Pakistan; Boissiere, Knight and Sabot (1985) for urban Kenya and Tanzania; Glewwe (1996) for Ghana; and Lavy, Spratt and Leboucher (1997) for Morocco.
- (2) on the impact of adult height on wages and/or productivity: Behrman and Deolalikar (1989) and Deolalikar (1988) for rural India; Haddad and Bouis (1991) for rural Philippines; Strauss (1986) for Côte d'Ivoire; Thomas and Strauss (1997) for Brazil; and Behrman (1993) for the more general experience in developing countries.
- (1) and (2) on the impact of early childhood nutrition and cognitive development on adult nutritional status and cognitive achievement: Grantham-McGregor et al. (1997) and Grantham-McGregor, Fernald, and Sethuraman (1999) for Jamaica; Martorell (1995) and Martorell, Rivera, and Kaplowitz (1989) for rural Guatemala; and Haas et al. (1996), Martorell (1999), and Martorell, Khan, and Schroeder (1994) for the more general experience in developing countries.
- (3) on the impact of grades of schooling completed on wages—hundreds of studies, many of which are surveyed in Psacharopoulos (1994) and Rosenzweig (1995);
- (3) and (4) on the impact of better child nutrition on progress through schooling: Jamison (1986) for China; Moock and Leslie (1986) for Nepal; and Behrman (1993), Leslie and Jamison (1990), and Pollitt (1990) for the more general experience in developing countries.

(4) on the impact of preschool child nutrition on age of starting school: Alderman et al. (2000) for rural Pakistan; Glewwe and Jacoby (1995) for Ghana; and Glewwe, Jacoby, and King (2000) for the Philippines.

Unfortunately, the PROGRESA data and analysis to date do not include direct estimates of these links. So for illustrative simulations we use estimates from Thomas and Strauss (1997) for (2). Strauss and Thomas (1998) analyze the relationship between adult earnings and height and completed years of schooling for male workers in another Latin American country, Brazil. They find that a 1 percent increase in height leads to a 2.4 percent increase in adult male earnings, in a regression of log hourly wages on height and completed grades of schooling, controlling for selectivity into employment. Our estimates in Section 5.1 imply that PROGRESA increases target children's height by about 1.2 percent. 16 Under the assumption that there is strong persistence of changes in small children's anthropometric development so that the percentage changes for adults equal those that we estimate for children, the impact from this effect alone would be a 2.9 percent increase in lifetime earnings. Under the assumption that there is less persistence of changes in small children's anthropometric development so that the percentage changes for adults equal half of those that we estimate for children, the impact from this effect alone would be a 1.4 percent increase in lifetime earnings. In addition to the effect through channel (2), there is evidence from the studies noted of significant positive

-

 $<sup>^{16}</sup>$  The mean height for children in the primary sample used was 80.0 centimeters in 1998 and 88.2 centimeters in 1999 and the estimated impact of PROGRESA is about 1.0 centimeter.

effects through the other channels. So these estimates, conditional on the extent of persistence from childhood to adults in anthropometric measures, probably are lower bounds on the full effects that would be obtained if all four channels were considered.

### 7. SUMMARY AND CONCLUSIONS

The nutrition of infants and small children is of considerable interest not only because of interest in the immediate welfare of those children, but also because their nutrition in this formative stage of life is widely perceived to have substantial persistent impact on their physical and mental development and on their health status as adults. Their physical and mental development, in turn, shapes their life-time options by affecting their schooling success and their post-schooling productivity. Improvements in the nutritional status of currently malnourished infants and small children, thus, potentially may have important payoffs over the long term. Within rural Mexico, stunting, or short height relative to standards established for healthy populations, is the major form of PEM. Wasting is much less of a problem.

One of the major components of the PROGRESA program has been directed toward improving the nutritional status of small children in poor rural communities in Mexico. Cross-sectional comparisons of health/height for children who received this PROGRESA treatment versus others who were in PROGRESA eligible households but who did not receive this treatment suggest no positive effect of PROGRESA either on child height on the average or on reducing the proportion of children who are stunted. But

these comparisons may be misleading because of the failure to control for unobserved child, parental and household, market and community characteristics that may be correlated with children receiving the PROGRESA treatment and because of the failure to control for systematic initial differences in that, on the average, the children in the control sample tended to have better anthropometric status than children in the treatment sample.

Our preferred estimates control for these factors, as well as secular trends and transitory fluctuations in prices and household consumption, by using fixed effects estimates. PROGRESA treatment is represented by those who reportedly received the nutritional supplements in the treatment group (less than 60 percent of those children in the treatment group) for children in the critical age range of 12–36 months. These estimates find significant impacts of receiving PROGRESA treatment in increasing child growth and in reducing the probability of child stunting. They imply an increase of about a sixth in mean growth per year for these children, which may be somewhat larger for children from poorer households and poorer communities but who come from households with more educated household heads. Furthermore, such estimates may be conservative because some children in the control localities have received these supplements; some households report receiving the supplement for more than 12 months; the supplement may not be fully consumed; and the INSP data indicate that in many households, the supplement is shared among family members.

However, even these conservative estimates may have important long-run consequences. Under the assumptions that (1) there is strong persistence of changes in

small children's anthropometric development so that the percentage changes for adults equal those (are half of those) that we estimate for children and (2) that adult anthropometric-earnings relations from elsewhere in Latin America apply to the labor markets in which these children will be working as adults, the impact from this effect alone would be a 2.9 percent (1.4 percent) increase in lifetime earnings. In addition, there are likely to be other effects through increased cognitive development, increased schooling, and lowered age of completing given levels of schooling through starting when younger and passing successfully grades at a higher rate. While these estimates of necessity are fairly speculative, they suggest that PROGRESA may be having fairly substantial effects on lifetime productivities and earnings of currently small children in poor households.

## REFERENCES

- Adair, L. 1999. Filipino children exhibit catch-up growth from age 2 to 12 years. *Journal of Nutrition* 129 (6): 1140–1148.
- Alderman, H. 1986. The effects of income and food price changes on the acquisition of food by low-income households. International Food Policy Research Institute, Washington, D.C.
- Alderman, H. 1993. New research on poverty and malnutrition: What are the implications for research and policy? In *Including the poor*, ed. M. Lipton and J. Van der Gaag. Washington, D.C.: World Bank.
- Alderman, H., J. R. Behrman, V. Lavy, and R. Menon. 2000. Child health and school enrollment: A longitudinal analysis. Washington, D.C.: World Bank. Photocopy.
- Alderman, H., J. R. Behrman, D. Ross, and R. Sabot. 1996. The returns to endogenous human capital in Pakistan's rural wage labour market. *Oxford Bulletin of Economics and Statistics* 58 (1): 29–56.
- Alderman, H., P.-A. Chiappori, L. Haddad, J. Hoddinott, and R. Kanbur. 1995. Unitary versus collective models of the household: Time to shift the burden of proof?

  World Bank Research Observer 10: 1–19.
- Becker, G. S. 1967. *Human capital and the personal distribution of income: An analytical approach*. Ann Arbor: University of Michigan, Woytinsky Lecture. Republished 1975. *Human Capital*. 2nd edition. New York: NBER, 94–117.

- Behrman, J. R. 1993. The economic rationale for investing in nutrition in developing countries. *World Development* 21 (11): 1749–1771.
- Behrman, J. R. 1996. The impact of health and nutrition on education. *World Bank Research Observer* 11: 23–37.
- Behrman, J. R. 1997. Intrahousehold distribution and the family. In *Handbook of population and family economics*, ed. M. R. Rosenzweig and O. Stark.Amsterdam: North-Holland Publishing Company.
- Behrman, J. R., and A. B. Deolalikar. 1987. Will developing country nutrition improve with income? A case study for rural South India. *Journal of Political Economy* 95 (3) (June): 108–138.
- Behrman, J. R., and A. B. Deolalikar. 1988. Health and nutrition. In *Handbook on economic development*, Vol. 1, ed. H. B. Chenery and T. N. Srinivasan.

  Amsterdam: North Holland Publishing Co.
- Behrman, J. R., and A. B. Deolalikar. 1989. Wages and labor supply in rural India: The role of health, nutrition and seasonality. In *Causes and implications of seasonal variability in household food security*, ed. D. E. Sahn. Baltimore, Md., U.S.A.: Johns Hopkins University Press.
- Behrman, J. R., and J. C. Knowles. 1999. Household income and child schooling in Vietnam. *World Bank Economic Review* 13 (2): 211–256.
- Behrman, J. R., and P. E. Todd. 1999. Randomness in the experimental samples of PROGRESA (education, health, and nutrition program). Report submitted to PROGRESA. International Food Policy Research Institute, Washington, D.C.

- Behrman, J. R., A. Foster, and M. R. Rosenzweig. 1997. The dynamics of agricultural production and the calorie-income relationship: Evidence from Pakistan. *Journal of Econometrics* 77 (1) (March): 187–207.
- Boissiere, M., J. B. Knight, and R. H. Sabot. 1985. Earnings, schooling, ability and cognitive skills. *American Economic Review* 75: 1016–1030.
- Bouis, H. E. 1994. The effect of income on demand for food in poor countries: Are our databases giving us reliable estimates? *Journal of Development Economics* 44 (1) (June): 199–226.
- Bouis, H. E., and L. J. Haddad. 1992. Are estimates of calorie-income elasticities too high? A recalibration of the plausible range. *Journal of Development Economics* 39 (2) (October): 333–364.
- Coady, D. P. 1999. Preliminary report on the application of social cost-benefit analysis to the evaluation of the human capital impacts of PROGRESA. Report submitted to PROGRESA. International Food Policy Research Institute, Washington, D.C.
- Deolalikar, A. B. 1988. Nutrition and labor productivity in agriculture: Estimates for rural South India. *Review of Economics and Statistics* 70 (3): 406–413.
- Glewwe, P. 1996. The relevance of standard estimates of rates of return to schooling for education policy: A critical assessment. *Journal of Development Economics* 51 (2): 267-290.
- Glewwe, P., and H. Jacoby. 1995. An economic analysis of delayed primary school enrollment and childhood malnutrition in a low-income country. *Review of Economics and Statistics* 77 (1): 156–169.

- Glewwe, P., H. Jacoby, and E. King. 2000. Early childhood nutrition and academic achievement: A longitudinal analysis. *Journal of Public Economics* (forthcoming).
- Golden, M. 1994. Is complete catch-up possible for stunted malnourished children? *European Journal of Clinical Nutrition* 48 (Supplement 1): S58–S71.
- Grantham-McGregor, S. L. Fernald, and K. Sethuraman. 1999. Effects of health and nutrition on cognitive and behavioural development in children in the first three years of life: Part 1: Low birth weight, breastfeeding and protein-energy malnutrition. *Food and Nutrition Bulletin* 20: 53–75.
- Grantham-McGregor, S., C. Walker, S. Chang, and C. Powell. 1997. Effects of early childhood supplementation with and without stimulation on later development in stunted Jamaican children. *American Journal of Clinical Nutrition* 66: 247–253.
- Haas, J, S. Murdoch, J. Rivera, and R. Martorell. 1996. Early nutrition and later physical work capacity. *Nutrition Reviews* 54: S41–S48.
- Haddad, L., and H. Bouis. 1991. The impact of nutritional status on agricultural productivity: Wage evidence from the Philippines. *Oxford Bulletin of Economics and Statistics* 53 (1): 45–68.
- Haddad, L., and J. Hoddinott. 1994. Women's income and boy-girl anthropometric status in the Côte d'Ivoire. *World Development* 22 (4) (April): 543–554.
- Haddad, L., J. Hoddinott, and H. Alderman, eds. 1997. *Intrahousehold resource allocation: Methods, models, and policy*. Baltimore, Md., U.S.A.: Johns Hopkins

  University Press for the International Food Policy Research Institute.

- Hoddinott, J., and L. Haddad. 1995. Does female income share influence household expenditures? Evidence from the Côte d'Ivoire. *Oxford Bulletin of Economics and Statistics* 57 (1) (February): 77–96.
- Hoddinott, J., and E. Skoufias. 2000. Preliminary evidence of the impact of PROGRESA on consumption. Report submitted to PROGRESA. International Food Policy Research Institute, Washington, D.C.
- INSP (Instituto Nacional de Salud Publica). 1998. Informe sobre prevalencias de desnutrición y amenia en la evaluación basal del PROGRESA (componente nutricion). Cuernavaca, Mexico.
- Jamison, D, T. 1986. Child malnutrition and school performance in China. *Journal of Development Economics* 20 (2): 299–310.
- Johnston, F., S. Low, Y. de Baessa, and R. MacVean. 1987. Interaction of nutritional and socioeconomic status as determinants of cognitive achievement in disadvantaged urban Guatemalan children. *American Journal of Physical Anthropology* 73: 501–506.
- Lavy, V., J. Spratt, and N. Leboucher. 1997. Patterns of incidence and change in Moroccan literacy. *Comparative Education Review* 41 (2).
- Leslie, J., and D. Jamison. 1990. Health and nutrition considerations in education planning 1. Educational consequences of health problems among school-age children. *Food and Nutrition Bulletin* 12: 191–203.
- Martorell, R. 1995. Results and implications of the INCAP follow-up study. *Journal of Nutrition* 125 (Supplement): 1127S–1138S.

- Martorell, R. 1999. The nature of child malnutrition and its long-term implications. *Food and Nutrition Bulletin* 20: 288–292.
- Martorell, R., K. L. Khan, and D. G. Schroeder. 1994. Reversibility of stunting:

  Epidemiological findings in children from developing countries. *European Journal of Clinical Nutrition* 48 (Supplement): S45–S57.
- Martorell, R., J. Rivera, and H. Kaplowitz. 1989. Consequences of stunting in early childhood for adult body size in rural Guatemala. Food Research Institute, Stanford University, Stanford, Calif., U.S.A. Photocopy.
- Moock, P. R., and J. Leslie. 1986. Childhood malnutrition and schooling in the Terai region of Nepal. *Journal of Development Economics* 20 (1): 33–52.
- Pollitt, E. 1990. Malnutrition and infection in the classroom. Paris: UNESCO.
- Pollitt, E., K. S. Gorman, P. L. Engle, R. Martorell, and J. Rivera. 1993. Early supplementary feeding and cognition. *Monographs of the Society for Research in Child Development* 235 (58): 7.
- Psacharopoulos, G. 1994. Returns to investment in education: A global update. *World Development* 22 (9): 1325–1344.
- Rivera, J., G. Rodríguez, T. Shamah, J. Rosado, E. Casanueva, I. Maulén, G. Toussaint, and A García-Aranda. 2000. Implementation, monitoring, and evaluation of the nutrition component of the Mexican Social Programme (PROGRESA). *Food and Nutrition Bulletin* 21 (1): 35–42.

- Rosado, J., J. Rivera, G. Lopez, L. Solano, G. Rodríguez, E. Casanueva, A. García-Aranda, G. Toussaint, and I. Maulén. 1999. Programa de suplementación para grupos con alto riesgo de desnutrición. *Salud Publica Mex* 41: 153–162.
- Rosenzweig, M. R. 1995. Why are there returns in schooling? *American Economic Review* 85 (2): 153–158.
- Skoufias, E., B. Davis, and J. R. Behrman. 1999. An evaluation of the selection of beneficiary households in the education, health, and nutrition program (PROGRESA) of Mexico. Report submitted to PROGRESA.: International Food Policy Research Institute, Washington, D.C.
- Strauss, J. 1986. Does better nutrition raise farm productivity? *Journal of Political Economy* 94: 297–320.
- Strauss, J., and D. Thomas. 1995. Human resources: Empirical modeling of household and family decisions. In *Handbook of development economics*, volume 3A, ed. J. R. Behrman and T. N. Srinivasan. Amsterdam: North-Holland Publishing Company.
- Strauss, J., and D. Thomas. 1998. Health, nutrition, and economic development. *Journal of Economic Literature* 36 (2): 766–817.
- Subramanian, S., and A. Deaton. 1996. The demand for food and calories. *Journal of Political Economy* 104 (1) (February).
- Thomas, D. 1990. Intrahousehold resource allocation: An inferential approach. *Journal of Human Resources* 25 (4) (Fall): 635–664.

- Thomas, D. 1993. The distribution of income and expenditure within the household.

  \*Annales de Economie et de Statistiques 29: 109–136.
- Thomas, D. 1994. Like father, like son; like mother, like daughter: Parental resources and child height. *Journal of Human Resources* 29 (4) (Fall): 950–989.
- Thomas, D., and J. Strauss. 1997. Health and wages: Evidence on men and women in urban Brazil. *Journal of Econometrics* 77 (1): 159–187.
- Young, M. E. 1995. *Investing in young children*. World Bank Discussion Paper No. 275. Washington, D.C.: World Bank.

- 01 Agricultural Technology and Food Policy to Combat Iron Deficiency in Developing Countries, Howarth E. Bouis, August 1994
- 02 Determinants of Credit Rationing: A Study of Informal Lenders and Formal Credit Groups in Madagascar, Manfred Zeller, October 1994
- 703 The Extended Family and Intrahousehold Allocation: Inheritance and Investments in Children in the Rural Philippines, Agnes R. Quisumbing, March 1995
- 04 Market Development and Food Demand in Rural China, Jikun Huang and Scott Rozelle, June 1995
- 05 Gender Differences in Agricultural Productivity: A Survey of Empirical Evidence, Agnes R. Quisumbing, July 1995
- Gender Differentials in Farm Productivity: Implications for Household Efficiency and Agricultural Policy,
   Harold Alderman, John Hoddinott, Lawrence Haddad, and Christopher Udry, August 1995
- 07 A Food Demand System Based on Demand for Characteristics: If There Is "Curvature" in the Slutsky Matrix, What Do the Curves Look Like and Why?, Howarth E. Bouis, December 1995
- 08 Measuring Food Insecurity: The Frequency and Severity of "Coping Strategies," Daniel G. Maxwell, December 1995
- 09 *Gender and Poverty: New Evidence from 10 Developing Countries*, Agnes R. Quisumbing, Lawrence Haddad, and Christine Peña, December 1995
- Women's Economic Advancement Through Agricultural Change: A Review of Donor Experience, Christine Peña, Patrick Webb, and Lawrence Haddad, February 1996
- Rural Financial Policies for Food Security of the Poor: Methodologies for a Multicountry Research Project, Manfred Zeller, Akhter Ahmed, Suresh Babu, Sumiter Broca, Aliou Diagne, and Manohar Sharma, April 1996
- 12 Child Development: Vulnerability and Resilience, Patrice L. Engle, Sarah Castle, and Purnima Menon, April 1996
- Determinants of Repayment Performance in Credit Groups: The Role of Program Design, Intra-Group Risk Pooling, and Social Cohesion in Madagascar, Manfred Zeller, May 1996
- 14 Demand for High-Value Secondary Crops in Developing Countries: The Case of Potatoes in Bangladesh and Pakistan, Howarth E. Bouis and Gregory Scott, May 1996
- 15 Repayment Performance in Group-Based credit Programs in Bangladesh: An Empirical Analysis, Manohar Sharma and Manfred Zeller, July 1996
- 16 How Can Safety Nets Do More with Less? General Issues with Some Evidence from Southern Africa, Lawrence Haddad and Manfred Zeller, July 1996
- 17 Remittances, Income Distribution, and Rural Asset Accumulation, Richard H. Adams, Jr., August 1996
- 18 Care and Nutrition: Concepts and Measurement, Patrice L. Engle, Purnima Menon, and Lawrence Haddad, August 1996
- 19 Food Security and Nutrition Implications of Intrahousehold Bias: A Review of Literature, Lawrence Haddad, Christine Peña, Chizuru Nishida, Agnes Quisumbing, and Alison Slack, September 1996
- 20 Macroeconomic Crises and Poverty Monitoring: A Case Study for India, Gaurav Datt and Martin Ravallion, November 1996

- 21 Livestock Income, Male/Female Animals, and Inequality in Rural Pakistan, Richard H. Adams, Jr., November 1996
- 22 Alternative Approaches to Locating the Food Insecure: Qualitative and Quantitative Evidence from South India, Kimberly Chung, Lawrence Haddad, Jayashree Ramakrishna, and Frank Riely, January 1997
- 23 Better Rich, or Better There? Grandparent Wealth, Coresidence, and Intrahousehold Allocation, Agnes R. Quisumbing, January 1997
- 24 Child Care Practices Associated with Positive and Negative Nutritional Outcomes for Children in Bangladesh: A Descriptive Analysis, Shubh K. Kumar Range, Ruchira Naved, and Saroj Bhattarai, February 1997
- 25 Water, Health, and Income: A Review, John Hoddinott, February 1997
- 26 Why Have Some Indian States Performed Better Than Others at Reducing Rural Poverty?, Gaurav Datt and Martin Ravallion, March 1997
- 27 "Bargaining" and Gender Relations: Within and Beyond the Household, Bina Agarwal, March 1997
- Developing a Research and Action Agenda for Examining Urbanization and Caregiving: Examples from Southern and Eastern Africa, Patrice L. Engle, Purnima Menon, James L. Garrett, and Alison Slack, April 1997
- 29 *Gender, Property Rights, and Natural Resources*, Ruth Meinzen-Dick, Lynn R. Brown, Hilary Sims Feldstein, and Agnes R. Quisumbing, May 1997
- 30 Plant Breeding: A Long-Term Strategy for the Control of Zinc Deficiency in Vulnerable Populations, Marie T. Ruel and Howarth E. Bouis, July 1997
- 31 Is There an Intrahousehold 'Flypaper Effect'? Evidence from a School Feeding Program, Hanan Jacoby, August 1997
- The Determinants of Demand for Micronutrients: An Analysis of Rural Households in Bangladesh, Howarth E. Bouis and Mary Jane G. Novenario-Reese, August 1997
- 33 Human Milk—An Invisible Food Resource, Anne Hatløy and Arne Oshaug, August 1997
- 34 The Impact of Changes in Common Property Resource Management on Intrahousehold Allocation, Philip Maggs and John Hoddinott, September 1997
- 35 Market Access by Smallholder Farmers in Malawi: Implications for Technology Adoption, Agricultural Productivity, and Crop Income, Manfred Zeller, Aliou Diagne, and Charles Mataya, September 1997
- 36 The GAPVU Cash Transfer Program in Mozambique: An assessment, Gaurav Datt, Ellen Payongayong, James L. Garrett, and Marie Ruel, October 1997
- Why Do Migrants Remit? An Analysis for the Dominican Sierra, Bénédicte de la Brière, Alain de Janvry, Sylvie Lambert, and Elisabeth Sadoulet, October 1997
- 38 Systematic Client Consultation in Development: The Case of Food Policy Research in Ghana, India, Kenya, and Mali, Suresh Chandra Babu, Lynn R. Brown, and Bonnie McClafferty, November 1997
- Whose Education Matters in the Determination of Household Income: Evidence from a Developing Country, Dean Jolliffe, November 1997
- 40 Can Qualitative and Quantitative Methods Serve Complementary Purposes for Policy Research? Evidence from Accra, Dan Maxwell, January 1998

- 41 The Political Economy of Urban Food Security in Sub-Saharan Africa, Dan Maxwell, February 1998
- 42 Farm Productivity and Rural Poverty in India, Gaurav Datt and Martin Ravallion, March 1998
- 43 How Reliable Are Group Informant Ratings? A Test of Food Security Rating in Honduras, Gilles Bergeron, Saul Sutkover Morris, and Juan Manuel Medina Banegas, April 1998
- 44 Can FAO's Measure of Chronic Undernourishment Be Strengthened?, Lisa C. Smith, with a Response by Logan Naiken, May 1998
- 45 Does Urban Agriculture Help Prevent Malnutrition? Evidence from Kampala, Daniel Maxwell, Carol Levin, and Joanne Csete, June 1998
- 46 Impact of Access to Credit on Income and Food Security in Malawi, Aliou Diagne, July 1998
- 47 Poverty in India and Indian States: An Update, Gaurav Datt, July 1998
- 48 *Human Capital, Productivity, and Labor Allocation in Rural Pakistan*, Marcel Fafchamps and Agnes R. Quisumbing, July 1998
- 49 A Profile of Poverty in Egypt: 1997, Gaurav Datt, Dean Jolliffe, and Manohar Sharma, August 1998.
- 50 Computational Tools for Poverty Measurement and Analysis, Gaurav Datt, October 1998
- 51 Urban Challenges to Food and Nutrition Security: A Review of Food Security, Health, and Caregiving in the Cities, Marie T. Ruel, James L. Garrett, Saul S. Morris, Daniel Maxwell, Arne Oshaug, Patrice Engle, Purnima Menon, Alison Slack, and Lawrence Haddad, October 1998
- 52 Testing Nash Bargaining Household Models With Time-Series Data, John Hoddinott and Christopher Adam, November 1998
- 53 Agricultural Wages and Food Prices in Egypt: A Governorate-Level Analysis for 1976-1993, Gaurav Datt and Jennifer Olmsted. November 1998
- 54 Endogeneity of Schooling in the Wage Function: Evidence from the Rural Philippines, John Maluccio, November 1998
- 55 Efficiency in Intrahousehold Resource Allocation, Marcel Fafchamps, December 1998
- 56 How Does the Human Rights Perspective Help to Shape the Food and Nutrition Policy Research Agenda?, Lawrence Haddad and Arne Oshaug, February 1999
- 57 The Structure of Wages During the Economic Transition in Romania, Emmanuel Skoufias, February 1999
- Women's Land Rights in the Transition to Individualized Ownership: Implications for the Management of Tree Resources in Western Ghana, Agnes Quisumbing, Ellen Payongayong, J. B. Aidoo, and Keijiro Otsuka, February 1999
- 59 Placement and Outreach of Group-Based Credit Organizations: The Cases of ASA, BRAC, and PROSHIKA in Bangladesh, Manohar Sharma and Manfred Zeller, March 1999
- 60 Explaining Child Malnutrition in Developing Countries: A Cross-Country Analysis, Lisa C. Smith and Lawrence Haddad, April 1999
- Does Geographic Targeting of Nutrition Interventions Make Sense in Cities? Evidence from Abidjan and Accra, Saul S. Morris, Carol Levin, Margaret Armar-Klemesu, Daniel Maxwell, and Marie T. Ruel, April 1999

- Good Care Practices Can Mitigate the Negative Effects of Poverty and Low Maternal Schooling on Children's Nutritional Status: Evidence from Accra, Marie T. Ruel, Carol E. Levin, Margaret Armar-Klemesu, Daniel Maxwell, and Saul S. Morris, April 1999
- 63 Are Urban Poverty and Undernutrition Growing? Some Newly Assembled Evidence, Lawrence Haddad, Marie T. Ruel, and James L. Garrett, April 1999
- 64 Some Urban Facts of Life: Implications for Research and Policy, Marie T. Ruel, Lawrence Haddad, and James L. Garrett, April 1999
- 65 Are Determinants of Rural and Urban Food Security and Nutritional Status Different? Some Insights from Mozambique, James L. Garrett and Marie T. Ruel, April 1999
- Working Women in an Urban Setting: Traders, Vendors, and Food Security in Accra, Carol E. Levin, Daniel G. Maxwell, Margaret Armar-Klemesu, Marie T. Ruel, Saul S. Morris, and Clement Ahiadeke, April 1999
- 67 Determinants of Household Access to and Participation in Formal and Informal Credit Markets in Malawi, Aliou Diagne, April 1999
- 68 Early Childhood Nutrition and Academic Achievement: A Longitudinal Analysis, Paul Glewwe, Hanan Jacoby, and Elizabeth King, May 1999
- 69 Supply Response of West African Agricultural Households: Implications of Intrahousehold Preference Heterogeneity, Lisa C. Smith and Jean-Paul Chavas, July 1999
- 70 Child Health Care Demand in a Developing Country: Unconditional Estimates from the Philippines, Kelly Hallman, August 1999
- 71 Social Capital and Income Generation in South Africa, 1993-98, John Maluccio, Lawrence Haddad, and Julian May, September 1999
- Validity of Rapid Estimates of Household Wealth and Income for Health Surveys in Rural Africa, Saul S. Morris, Calogero Carletto, John Hoddinott, and Luc J. M. Christiaensen, October 1999
- 73 Social Roles, Human Capital, and the Intrahousehold Division of Labor: Evidence from Pakistan, Marcel Fafchamps and Agnes R. Quisumbing, October 1999
- 74 Can Cash Transfer Programs Work in Resource-Poor Countries? The Experience in Mozambique, Jan W. Low, James L. Garrett, and Vitória Ginja, October 1999
- 75 Determinants of Poverty in Egypt, 1997, Gaurav Datt and Dean Jolliffe, October 1999
- 76 Raising Primary School Enrolment in Developing Countries: The Relative Importance of Supply and Demand, Sudhanshu Handa, November 1999
- 77 The Political Economy of Food Subsidy Reform in Egypt, Tammi Gutner, November 1999.
- 78 Determinants of Poverty in Mozambique: 1996-97, Gaurav Datt, Kenneth Simler, Sanjukta Mukherjee, and Gabriel Dava, January 2000
- 79 Adult Health in the Time of Drought, John Hoddinott and Bill Kinsey, January 2000
- 80 Nontraditional Crops and Land Accumulation Among Guatemalan Smallholders: Is the Impact Sustainable? Calogero Carletto, February 2000
- 81 The Constraints to Good Child Care Practices in Accra: Implications for Programs, Margaret Armar-Klemesu, Marie T. Ruel, Daniel G. Maxwell, Carol E. Levin, and Saul S. Morris, February 2000

- Pathways of Rural Development in Madagascar: An Empirical Investigation of the Critical Triangle of Environmental Sustainability, Economic Growth, and Poverty Alleviation, Manfred Zeller, Cécile Lapenu, Bart Minten, Eliane Ralison, Désiré Randrianaivo, and Claude Randrianarisoa, March 2000
- 83 Quality or Quantity? The Supply-Side Determinants of Primary Schooling in Rural Mozambique, Sudhanshu Handa and Kenneth R. Simler, March 2000
- 84 Intrahousehold Allocation and Gender Relations: New Empirical Evidence from Four Developing Countries, Agnes R. Quisumbing and John A. Maluccio, April 2000
- 85 Intrahousehold Impact of Transfer of Modern Agricultural Technology: A Gender Perspective, Ruchira Tabassum Naved, April 2000
- Women's Assets and Intrahousehold Allocation in Rural Bangladesh: Testing Measures of Bargaining Power, Agnes R. Quisumbing and Bénédicte de la Brière, April 2000
- 87 Changes in Intrahousehold Labor Allocation to Environmental Goods Collection: A Case Study from Rural Nepal, Priscilla A. Cooke, May 2000
- 88 The Determinants of Employment Status in Egypt, Ragui Assaad, Fatma El-Hamidi, and Akhter U. Ahmed, June 2000
- 89 The Role of the State in Promoting Microfinance Institutions, Cécile Lapenu, June 2000
- 90 Empirical Measurements of Households' Access to Credit and Credit Constraints in Developing Countries: Methodological Issues and Evidence, Aliou Diagne, Manfred Zeller, and Manohar Sharma, July 2000
- 91 Comparing Village Characteristics Derived From Rapid Appraisals and Household Surveys: A Tale From Northern Mali, Luc Christiaensen, John Hoddinott, and Gilles Bergeron, July 2000
- 92 Assessing the Potential for Food-Based Strategies to Reduce Vitamin A and Iron Deficiencies: A Review of Recent Evidence, Marie T. Ruel and Carol E. Levin, July 2000
- 93 Mother-Father Resource Control, Marriage Payments, and Girl-Boy Health in Rural Bangladesh, Kelly Hallman, September 2000
- 94 Targeting Urban Malnutrition: A Multicity Analysis of the Spatial Distribution of Childhood Nutritional Status, Saul Sutkover Morris, September 2000
- 95 Attrition in the Kwazulu Natal Income Dynamics Study 1993-1998, John Maluccio, October 2000
- 96 Attrition in Longitudinal Household Survey Data: Some Tests for Three Developing-Country Samples, Harold Alderman, Jere R. Behrman, Hans-Peter Kohler, John A. Maluccio, Susan Cotts Watkins, October 2000
- 97 Socioeconomic Differentials in Child Stunting Are Consistently Larger in Urban Than in Rural Areas, Purnima Menon, Marie T. Ruel, and Saul S. Morris, December 2000
- 98 Participation and Poverty Reduction: Issues, Theory, and New Evidence from South Africa, John Hoddinott, Michelle Adato, Tim Besley, and Lawrence Haddad, January 2001
- 99 *Cash Transfer Programs with Income Multipliers: PROCAMPO in Mexico*, Elisabeth Sadoulet, Alain de Janvry, and Benjamin Davis, January 2001
- 100 On the Targeting and Redistributive Efficiencies of Alternative Transfer Instruments, David Coady and Emmanuel Skoufias, March 2001

